



Article An Integrated Framework for Estimating Origins and Destinations of Multimodal Multi-Commodity Import and Export Flows Using Multisource Data

Muhammad Safdar ^{1,2,3}, Ming Zhong ^{1,2,3,4,*}, Zhi Ren ^{1,2,3} and John Douglas Hunt ⁵

- ¹ Intelligent Transportation Systems Research Center, Wuhan University of Technology, Wuhan 430063, China; safdar@whut.edu.cn (M.S.); renzhi@whut.edu.cn (Z.R.)
- ² National Engineering Research Center for Water Transport Safety, Wuhan University of Technology, Wuhan 430063, China
- ³ State Key Laboratory of Maritime Technology and Safety, Wuhan University of Technology, Wuhan 430063, China
- ⁴ Department of Civil and Environmental Engineering, University of Waterloo, Waterloo, ON N2L 3G1, Canada
- ⁵ Department of Civil Engineering, University of Calgary, 2500 University Drive NW, Calgary, AB T2N 1N4, Canada
- * Correspondence: mzhong@whut.edu.cn

Abstract: Estimating origin-destination (OD) demand is integral to urban, regional, and national freight transportation planning and modeling systems. However, in developing countries, existing studies reveal significant inconsistencies between OD estimates for domestic and import/export commodities derived from interregional input-output (IO) tables and those from regional IO tables. These discrepancies create a significant challenge for properly forecasting the freight demand of regional/interregional multimodal transportation networks. To this end, this study proposes a novel integrated framework for estimating regional and international (import/export) OD freight flows for a set of key commodities that dominate long-distance transportation. The framework leverages multisource data and follows a three-step process. First, a spatial economic model, PECAS activity allocation, is developed to estimate freight OD demand within a specific region. Second, the international (import and export) freight OD is estimated from different zones to foreign countries, including major import and export nodes such as international seaports, using a gravity model with the zone-pair friction obtained from a multimodal transportation model. Third, the OD matrices are converted from monetary value to tonnage and assigned to the multimodal transportation super network using the incremental freight assignment method. The model is calibrated using traffic counts of the highways, railways, and port throughput data. The proposed framework is tested through a case study of the Province of Jiangxi, which is crucial for forecasting freight demand before the planning, design, and operation of the Ganyue Canal. The predictive analytics of the proposed framework demonstrated high validity, where the goodness-of-fit (R^2) between the observed and estimated freight flows on specific links for each of the three transport modes was higher than 0.9. This indirectly confirms the efficacy of the model in predicting freight OD demands. The proposed framework is adaptable to other regions and aids practitioners in providing a comprehensive tool for informed decision-making in freight demand modeling.

Keywords: origin-destination matrices; freight demand modeling; multimodal transportation; commodities; PECAS; input-output table; integrated modeling

1. Introduction

The accurate freight estimation of origin and destination (OD) is paramount for effective transportation planning and demand modeling at the national, state, and local levels [1,2]. Understanding freight movement within complex multimodal networks helps



Citation: Safdar, M.; Zhong, M.; Ren, Z.; Hunt, J.D. An Integrated Framework for Estimating Origins and Destinations of Multimodal Multi-Commodity Import and Export Flows Using Multisource Data. *Systems* 2024, *12*, 406. https:// doi.org/10.3390/systems12100406

Academic Editors: Shuqi Xue, Yun Wang and Xiaomeng Shi

Received: 17 August 2024 Revised: 24 September 2024 Accepted: 25 September 2024 Published: 30 September 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). planners anticipate and manage flows while facilitating the vital growth of economic activities without overburdening existing infrastructure [2-4]. As freight volumes continue to increase, transportation systems face mounting pressure. This calls for more sophisticated methods to track, monitor, and analyze commodity movements, as well as to evaluate their impacts on transportation nodes and networks. Traditional freight demand models, such as direct facility flow factoring, OD factoring, truck models, four-step commodity models, and economic activity models, have been widely used. Commodity-based models, in particular, are valuable for analyzing supply and demand within economic sectors [5–7]. However, these models often face limitations at the regional and international levels, primarily due to data constraints. Consequently, they may not adequately capture the complexity of freight flows across multimodal transportation systems [7,8]. Recent advancements in freight modeling have shifted from traditional regression-based methods to more integrated approaches. These newer methods combine multiple data sources and modeling techniques, offering a detailed understanding of freight flows. Freight OD estimation is closely linked to economic activity and reflects the demand for raw materials, intermediate goods, and finished products across regions [7]. However, freight flow estimation faces challenges related to data availability and methodological approaches, where the appropriate OD matrix estimation model depends largely on the quality of data and the scale of the study area [7–10].

Establishing interregional connections in a Multi-Regional Input-Output (MRIO) analysis is fundamental. In practice, freight OD estimation employs both survey- and non-survey-based methods. While, survey-based methods, particularly for interregional trade flows within a country, are not always available because of the significant workload and financial constraints, making these methods time-consuming, labor-intensive, and expensive [11–13]. Therefore, non-survey methods such as gravity- and behavior-based models were employed [14]. The non-survey approach is more practical and facilitates the operationalization of regional input–output (IO) data estimation [15]. The existing literature on regional and interregional freight OD estimation underscores a variety of methodologies yet reveals persistent gaps that require further exploration. For instance, gravity models have been extensively employed because of their simplicity in estimating interregional trade flows, as evidenced by Ramos and Sargento [16] and Lindall et al. [17]. These models are particularly helpful in situations where data are incomplete, however, they often require additional information to refine parameters, such as distance decay. Moreover, the integration of economic and transportation data via methods such as MRIO analysis have been explored to estimate regional trade flows [18,19]. Nevertheless, challenges arise in reconciling these estimates with provincial data. This issue is particularly pronounced in developing countries, where discrepancies hinder the accurate forecasting of freight demand and efficient planning of multimodal networks [20].

In the case of China, provincial and interregional IO tables released by the National Statistics Bureau often exhibit inconsistencies in trade flow data, such as mismatches between domestic inflows and outflows across provinces [20-22]. Moreover, these IO tables lack sufficient detail regarding the movement of specific commodities across provinces and international borders, providing only aggregate data on domestic flows and limited information on foreign imports and exports [20,23]. Many existing freight models focus on specific planning units or commodity groups, neglect adjacent regions and critical international trade corridors, and fail to account for the interconnected nature of modern logistics networks, where freight flows in one region can have significant downstream effects on industries in neighboring regions or international markets [20,24,25]. Thus, there is an urgent need for a modeling framework that integrates both regional and interregional trade data, accounting for the broader context in which provinces and the country function [26]. A model that links seaports to domestic-foreign OD flows is particularly important for spatial interaction modeling [6,25]. To address these gaps, this study proposes a novel integrated framework for estimating freight flows in intra-regional and international (import and export) contexts, linked through a two-tier multimodal transportation super network. This approach improves the accuracy of OD flow estimates by incorporating multisource data, including provincial and interregional IO tables, customs data, land use data, transportation networks, truck GPS data, railway waybills, ship visas, and statistical yearbooks. Consequently, it provides a more comprehensive estimation of OD matrices, particularly for the long-distance transportation of key commodities.

The specific contributions of this study are threefold. First, a spatial economic model, the Production, Exchange, and Consumption Allocation System (PECAS) activity allocation is developed to estimate freight OD within a specific region. Second, the international (import and export) freight OD is estimated from different zones to foreign countries, including major import and export nodes such as international seaports, using a gravity model with zone-pair friction obtained from a multimodal transportation model. Third, the OD matrices are converted from monetary value to tonnage and assigned to the multimodal transportation super network using the incremental freight assignment method. The model is calibrated using traffic counts of highways, railways, and port throughput data. Additionally, the calibration of the trip length distribution for extended regions is achieved by integrating data from truck GPS, railway waybills, and ship visas. The proposed framework is tested through a case study of the Province of Jiangxi, which is crucial for forecasting freight demand before the planning, design, and operation of the Ganyue Canal. The proposed framework is adaptable to other regional areas and provides a useful tool that helps practitioners use recent multisource data to make better and more informed decisions.

The remainder of this paper is organized as follows. Section 2 reviews the relevant literature. Section 3 describes the methodology used in the study. Section 4 describes the implementation and testing of the proposed integrated model through a case study. Section 5 presents the results and analysis of freight OD flows and multimodal freight assignments. Finally, Section 6 concludes the study with the main findings, recommendations, implications, and avenues for future research.

2. Literature Review

Estimating OD is an inevitable step in passenger and freight demand modeling, impact analysis, and assessment studies, which necessitates the development of OD matrices [27–30]. Although gathering information for OD matrices entails considerable effort and costs in primary data collection, proprietary concerns often limit data sharing by freight companies [7,31]. The literature on freight OD estimation diverges between the structured and unstructured approaches. Structured studies typically employ closedform distribution models such as the gravity model, which focuses on optimizing the model fit and tailoring it to specific commodities, as seen in the works of Holguín-Veras and Patil [32], and Levine et al. [4], who expanded these models to include truck and rail modes for U.S. container flow estimation. Conversely, unstructured approaches, as discussed by Kalahasthi et al. [33], do not adhere to specific functional forms for trip distribution and often rely on existing data such as partial OD matrices. These methods utilize various techniques including entropy maximization, artificial neural networks, and maximum likelihood. Al-Battaineh and Kaysi [34], and Ma et al. [35], used a genetic algorithm and Bayesian networks, respectively, for truck OD estimation. In addition, Teye and Hensher [36], and Hensher et al. [37], developed new entropy maximization models for freight trip distribution in Australia. Holguín-Veras et al. [38] estimated commodity flow OD matrices based on observed traffic counts. Typically, freight truck demand modeling relies on OD matrices derived from expanded roadside or truck survey data, representing steady-state scalability for future predictions. However, this approach has limitations, notably, the lack of policy responsiveness. The aggregate four-steps model (FSM) is also prevalent for truck trip estimation [39], and conventional travel demand models fail to capture the prevalent trip-chaining behavior in truck trips. In parallel, there has been a growing inclination towards micro-level activity models, which provide detailed insights but are constrained by extensive data requirements [39,40]. This challenge has led to the

4 of 35

development of new methodologies that balance data availability with accurate forecasting. For instance, Pan et al. [1] leveraged machine learning and passively collected data to estimate multimodal OD matrices. Similarly, Basso et al. [41] used toll collection data and GPS information to estimate truck OD matrices using a multisource approach and decision tree modeling.

In addition, Holguin-Veras and Thorson [9] utilized a trip length distribution (TLD) for freight demand in Guatemala using survey data to analyze the spatial patterns of freight movements, similar to the use of TLD in passenger transportation, to represent the frequency of trips over varying distances. Their study found that freight generators significantly influenced TLD patterns, and that the shape of the distribution varied depending on the type of movement involved. Notably, Tavasszy and Stada [42], and Al-Battaineh and Kaysi [34], incorporated IO formulations with genetic algorithms for OD matrix estimation, thereby demonstrating the integration of economic and transportation data. By further expanding the horizon, Bachmann et al. [18] introduced a method to estimate regional trade flows using transportation survey data, converting freight OD flows into production-consumption trade flows. This method reconciles trade flow estimates with regional production and consumption, adhering to MRIO accounting principles. Integrated land-use and transportation models have become increasingly valuable for modeling freight demand within a broader spatial and economic context. A prominent example is the PECAS-based Statewide Integrated Model (SWIM) used in Oregon, USA (Versions 2 and 2.5), and the PECAS-based freight model applied in Alberta, Canada [43,44]. PECAS has also been employed in developing statewide transportation and land-use systems for Ohio and Oregon, along with urban land-use models in cities such as Calgary, Edmonton, Caracas, Wuhan, Shanghai, Guangzhou, and Mumbai [45]. The Quick Response Freight Manual of the US department of transportation, used in the Phoenix Metropolitan Area, estimates truck traffic based on employment-related trip-generation rates. However, its default parameters may not be easily transferable between regions, and its focus on vehicle trips rather than commodities may not capture the underlying economic supply and demand dynamics [8].

In the realm of interregional trade flows, gravity models have received significant attention because of their simplicity and effectiveness in estimating such flows even with missing data [46,47]. Ramos and Sargento [16], Sargento et al. [14], Lindall et al. [17], and Nakano and Nishimura [48] exemplify the utility of the gravity model for estimating commodity flows. Moreover, the development of interregional social accounting matrices for comprehensive regional analysis, as undertaken by Jackson et al. [49] using IMPLAN and commodity flow data from the U.S. Bureau of Transportation Statistics, illustrates the integration of detailed economic and transportation data to enhance the understanding of regional dynamics. Park et al. [19] presented a two-step approach to estimate US trade tables between all 50 states plus Washington, D.C., and the rest of the world as the basis for a new MRIO model. Sargento et al. [14] studied three different interregional trade estimation methodologies: a gravity model with an arbitrarily set distance decay parameter, a proportional flow model, and a gravity model with an estimated distance decay parameter. They found that a gravity-based model with an estimated distance decay parameter yielded the most accurate results.

Recent research has made strides to improve multimodal freight OD estimation techniques. Kalahasthi et al. [33] proposed a synthesis model that integrates trip distribution, mode choice, and empty trip estimation for freight. Osorio-Mora, et al. [50] introduced a mixed-integer linear programming model for multimodal, multi-commodity hub location problems by incorporating OD flows as decision variables. Uddin, et al. [51] developed a stochastic freight traffic assignment model that accounts for uncertainties in large-scale intermodal networks caused by disruptions such as natural disasters. Thoen, et al. [52] developed a shipment-based model for tour formation, using an iterative algorithm with incremental allocation of shipments. Yamaguchi et al. [53] presented a global logistics intermodal network simulation model that incorporated land transport submodules using incremental assignment methods. Xia and Ma [54] introduced a multimodal multi-commodity transportation network model for the strategic planning of freight flows, focusing on spatial price equilibrium and network equilibrium. Jain et al. [55] explored perspectives on rail-truck multimodal freight collaboration and online exchange platforms for matching supply and demand in multimodal services. Pompigna and Mauro [56] developed macro-level IO models for freight transport demand analysis, linking the flow of goods transported along a multimodal corridor at the broader economic system.

In summary, existing freight demand studies often focus on a single province or urban areas, specific commodities (such as cereal grains), or multimodal transportation models restricted to individual planning units, frequently overlooking adjacent regions and international trade corridors [7,25]. This represents a critical gap in the literature, as provinces operate within a broader, interconnected logistics network, where the production and flow of goods in one region can be vital to industries in another. Neglecting these external interactions may lead to underestimating freight demand, potentially undermining transportation planning and infrastructure development strategies [6,25]. To address this, this study proposes a novel framework for estimating intra-regional and international commodity flows (imports and exports), linked through a two-tier multimodal transportation super network. This framework aims to estimate regional and interregional OD flows more accurately for the key commodities involved in long-distance transport. By leveraging big data technology, this integrated approach offers a robust tool for generating precise freight OD estimates and improving the understanding of complex flow patterns.

3. Methodology

The development of sophisticated freight demand models remains a challenging task worldwide due to the scarcity of data, theoretical frameworks, and specialized expertise [57]. Freight modeling often resorts to simplistic methods, such as assuming that freight trips follow the same behavioral patterns as passenger trips, which overlooks key differences in the movement patterns and economic drivers of freight transportation [5,6]. Moreover, existing time-series models fail to capture spatial interactions, which are crucial for freight movement across regions [58]. Current approaches, including FSM, often simulate transportation and land-use in isolation, resulting in a static representation that does not reflect the dynamic interplay between transportation systems and broader economic activities [45]. Although logistics and tour-based models have improved accuracy at the local level, they struggle to predict flows at the network level, limiting their applicability in multiscale contexts [8]. In addition, tour-based models rely on survey data, which can be expensive and labor-intensive, thus presenting challenges for their widespread adoption. A review of the literature indicates that no single model can fulfill all the objectives of freight OD demand [5,58]. Each model has its specific strengths and limitations, making the selection of a model crucial, depending on the context, availability, quality of data, and goals of the study [10].

Recent advancements in freight modeling have shifted towards integrated approaches, which offer a more nuanced understanding of freight flows by combining multiple data sources and modeling techniques [45,59]. Therefore, this study proposes an integrated framework to estimate freight flows at multiple scales, including regional and international (imports and exports) levels using multisource data. The primary research steps are as follows:

Step 1: Estimate Intra-regional Commodity-Specific OD

The PECAS activity allocation (PECAS AA) model is used to estimate the OD matrices at the Traffic Analysis Zone (TAZ) level, which corresponds to the counties and districts within a region. The PECAS AA model can be understood as the aggregate result of a joint choice model involving decisions regarding location, production, consumption, and trade destinations for households, businesses, and other institutions. This approach aligns with the principles of micro-simulation for both businesses and households. Additionally, if prices are adjusted over time in response to excess demand (rather than forcing excess demand to zero within each period), the PECAS AA model offers a dynamic representation of the economy consistent with Walrasian economic theory [60]. In this way, PECAS serves as a practical progression toward fully dynamic aggregate models and dynamic micro-simulation models.

Unlike traditional models, the PECAS AA model integrates spatial IO tables with land use and transportation linkages, providing a more accurate spatial representation of freight demand. Drawing from models such as MEPLAN [61], TRANUS [62], DELTA [63], and UrbanSim [64], PECAS enables the integrated and interconnected allocation of activity quantities based on spatial proximity and accessibility. This integration allows for a detailed spatial analysis, providing insights into the location of production, and exchange and consumption activities, which are often oversimplified in traditional models. Thus, PECAS AA offers a sophisticated method for estimating commodity-specific freight flows within a region [65].

Step 2: Estimate international Commodity-Specific OD

To estimate the commodity OD for international imports and exports from provinces to foreign countries, including international seaport entry–exit points, a gravity model with the zone-pair friction obtained from a multimodal transportation model was used. Gravity models are widely used to estimate interregional trade flows by assuming that trade volumes are proportional to production at the origin, consumption at the destination, and the generalized cost of transportation or other forms of impedance between regions. This method is particularly effective for estimating trade flows from provinces to foreign countries, including international seaport entry–exit points [46].

Step 3: Load OD Matrices onto Multimodal Transportation Network

The OD matrices obtained from Steps 1 and 2 are converted from monetary value to tonnage, and assigned to the multimodal transportation supernetwork using the incremental freight assignment method. The model is calibrated using the traffic counts of highways, railways, and port throughput data. In addition, calibration of the trip length distribution for extended regions is achieved by integrating big data sources, including trucks' GPS data, railways waybills, and ship visas. The adoption of big data in transport and logistics sectors can lead to operational improvements and new business values [66].

Data from multiple sources are utilized to estimate both regional and international import/export freight flows. Intra-regional OD estimates are conducted at TAZ levels (district and county), and international import/export estimates are derived from regional zones to international seaport zones. The control totals are used as checks at various points during the process to ensure accuracy and consistency. The PECAS software package (licensed under the Apache License version 2.0.) (https://www.hbaspecto.com/, accessed on 23 September 2024) is used for the activity allocation model, and its results are linked to the Cube Voyage software (Version 6.5.1) (https://www.bentley.com/software/openpaths, accessed on 23 September 2024) for multimodal transportation super network simulation. Additional tools including ESRI's ArcGIS 10.8 for spatial analysis (www.esri.com, accessed on 23 September 2024), Python 3.8 (https://docs.python.org/3.8/, accessed on 23 September 2024), and Microsoft Excel (Microsoft 365) for data processing are used. The overall framework is illustrated in Figure 1, which provides a schematic overview of the proposed integrated approach.



Figure 1. Workflow of the integrated framework for estimation of freight OD flows of regional and international import and export flows.

4. Case Study

This study applies the proposed integrated framework to Jiangxi Province as a case study. Owing to globalization, economic reforms, and the opening-up policy, import and export activities have significantly increased at major ports in Mainland China, particularly in Jiangxi Province. In 2005, Jiangxi's GDP was CNY 394.1 billion, with imports and exports totaling CNY 40.7 billion. By 2019, GDP soared to CNY 2.4758 trillion, and the value of imports and exports reached CNY 508.9 billion [67]. These figures highlight the province's continuous economic growth and expanding international trade during the study period, which can be attributed to Jiangxi's strategic location, resource abundance, and diverse industries [24]. Leveraging its economic and geographical significance, the Chinese government initiated the Ganyue Canal or Jiangxi-Guangdong Canal project, spanning 1300 km (with 800 km in Jiangxi province and 500 km in Guangdong province). The Ganyue canal connects two major rivers: the Ganjiang River in Jiangxi and the Beijiang River in Guangdong, with an estimated investment of over CNY 150 billion (21 billion USD) [11,68]. The Ganyue Canal is expected to boost import and export activities in Guangdong and Fujian, which host international waterway ports and extend benefits to Hubei and Hunan provinces, potentially impacting the entire Yangtze River Basin, Pearl River, and global trade [68]. Thus, accurate OD freight distribution is essential to ensure seamless goods flow across interregional and international borders, which is vital for global economic growth, trade, and regional integration.

4.1. Data Sources and Preprocessing

The PECAS AA model for Jiangxi Province relies on multiple datasets, with the primary input derived from the aggregated economic flow (AEF) table, which is an extension of IO tables. These tables reflect the production and consumption relationships among various activities across different commodities. For commodities, the production and consumption quantities were directly obtained from the IO table. Labor production quantities were estimated using the number of households from the Jiangxi Statistical Yearbook, and the labor compensation data from the IO table. Labor consumption, on the other hand, was allocated to each activity based on labor compensation values from the same IO tables [69]. Space consumption quantities were calculated using employment data and space-use coefficients sourced from related studies [67], and were estimated using remote sensing data.

To ensure the consistency and accuracy of the input data for the PECAS AA model, a rigorous data preprocessing protocol was followed. In Jiangxi Province, two versions of IO tables are available—one with 139 activities and another with 42 activities. We adopted the 42-activity format for its simplicity, while certain sectors were further subdivided based on a more detailed 139-activity table to align with the specific requirements of the PECAS AA model. Sectors corresponding directly to the 42-activity format were used as-is, while others were subdivided based on weights derived from the 139-activity format. This study addressed data anomalies including missing values, outliers, and extreme observations. To mitigate these issues, this study employed exogenous factors and statistical imputation methods [69]. Specifically, GDP growth rates and total output values of production of corresponding activities from the IO table were utilized to align the data with the original trend of activities in Jiangxi province.

Employment data pose formatting challenges, particularly with duplicated county/districts names and administrative codes. These discrepancies were corrected to match the PECAS AA data format and Cube Voyage software requirements. Similarly, the TAZs numbering in the PECAS activity constraints and floorspace files were standardized to align with the TAZ shapefile of the study area, ensuring smooth model execution and avoiding complications.

Transportation network data were obtained from OpenStreetMap (https://planet. openstreetmap.org, accessed on 23 September 2024), however, the extracted network data initially contained discrepancies such as misaligned nodes and links, along with duplicated and dangling links. These issues were addressed through a comprehensive data cleaning process in Cube Voyage and GIS software (ArcGIS 10.8), during which incorrect links were removed, and the actual network structure was reconstructed. Additionally, centroid connectors, which are necessary to link TAZs to the transportation network, were generated using Cube Voyage software as they were not provided by OpenStreetMap.

- (1) Economic Data: Economic data for Jiangxi Province were obtained from multiple sources. These include the Jiangxi Statistical Yearbook (2010–2020) and regional IO tables (2012–2017), available from the Jiangxi Statistical Department (http://tjj.jiangxi.gov.cn/col/col38595/index.html, accessed on 23 September 2024). Additionally, the Interregional IO tables for 2012 were sourced from China's National Statistics Bureau (https://data.stats.gov.cn/, accessed on 23 September 2024), along with demographic data from the 2012 national population census. Employment survey data for 2012 were obtained from China's National Data (https://data.stats.gov.cn/english/index.htm, accessed on 23 September 2024), and the China Labor Statistical Yearbook (2010–2020) (https://www.chinayearbooks.com/tags/china-labour-statistical-yearbook, accessed on 23 September 2024). The regional macroeconomic data incorporate data on population, households, and employment, with regional IO tables compiled for China's 31 provinces based on the 2012 base year (http://www.shujuku.org/, accessed on 23 September 2024).
- (2) Land Use Data: The land use data for Jiangxi Province models incorporating space price data, TAZs, land regulatory data, floor space data, space use coefficients, floor area ratios [67], and remote sensing data were sourced from the Institute of Geographic Sciences and Natural Resources Research of the Chinese Academy of Sciences resources (http://www.igsnrr.cas.cn, accessed on 23 September 2024), and China's environmental data (http://www.resdc.cn/Datalist1.aspx, accessed on 23 September 2024).
- (3) Multimodal Transportation Data: Data for developing multimodal transportation in Jiangxi Province and extended region models were sourced from the China Transportation Statistical Yearbook 2012, the China Customs Statistical Yearbook 2012, and datasets on highways, railways, and waterway networks. These data include highway traffic surveys, railway freight volumes, and port inflow and outflow data. The highway network data covering various road types and tolls were extracted from OpenStreetMap (https://planet.openstreetmap.org, accessed on 23 September 2024). Waterway network data, including navigable waterways, locks, and ports, along with railway network data consisting of railway lines and stations, were sourced from the ArcGIS Online database. This study also integrated cargo reloading times, personnel salaries, freight rates, and monetary values from a multimodal transportation model, including the unit weight values of goods and transport-related socioeconomic activities from the China Customs Statistical Yearbook (http://www.customs.gov.cn/ customs/302249/zfxxgk/2799825/302274/tjfwzn/2319672/index.html, accessed on 23 September 2024). Additionally, it encompasses vessel traffic flow, freight data, customs import/export data, port data, truck GPS, railway waybills, ship visas, and navigation-lock data.

4.2. Intra-Regional Commodity OD

In the intra-regional model, the Jiangxi region was selected as a case study and divided into 99 TAZs, corresponding to the province's district and county boundaries (Figure 2). These TAZs were used to estimate the intra-regional freight OD demand using the PECAS AA model [65]. The PECAS AA model identifies optimal activity locations, technologies used, purchasing inputs, and selling outputs under equilibrium conditions. It employed a three-level nested logit model based on the stochastic utility-based framework developed by Hunt and Abraham [65]. The highest layer distributes regional socioeconomic activities across TAZs; the middle layer allocates specific activities to intra-regional zones based on technology patterns; and the lowest layer assigns commodities and services to exchange locations using two logit models where they are bought and sold. The PECAS AA model

models exchange zones that serve as markets where commodity supply and demand interact and are influenced by price and transportation impedance.

 Highest Level: The highest level allocates the overall activity quantities across specific regional zones (Equations (1) and (2)).

$$W_{a,z} = T_{a\zeta} \cdot \left(\frac{\exp(\lambda_{1,a} \cdot LU_{a,z})}{\sum_{z \in Z} \exp(\lambda_{1,a} \cdot LU_{a,z})} \right)$$
(1)

where:

$$LU_{a,z} = a_{\text{size},a} \cdot \frac{1}{\lambda_{1,a}} \cdot \ln(\text{Size}_{a,z}) + a_{inert,a} \cdot \ln(\text{PrevW}_{a,z} + \text{InertCons}_{a}) + \text{Const}_{a,z} + \sum_{v \in V} (a_{a,v} \cdot X_{v,z}) + a_{\text{tech},a} \cdot \text{CUTech}_{a,z}$$
(2)

where the location utility for a unit of activity *a* in zone $z(LU_{a,z})$ is calculated as a combination of various factors, including the size term $(\alpha_{\text{size},a} \cdot \frac{1}{\lambda_{1,a}} \cdot \ln(\text{Size}_{a,z}))$ accounts for the influence of the zone's size, $\lambda_{1,a}$ is the utility function dispersion parameter for allocation of activity a among location zones, and $\text{Size}_{a,z}$ represents relative size of zone *z* for activity a, the previous quantity term $(\alpha_{\text{inert},a} \cdot \ln(\text{Prev } W_{a,z} + \text{InertCons}_a))$, a constant specific to the zone (Const $\text{Con}_{a,z}$), with $\alpha_{\text{inert},a}$ representing the sensitivity to the previous amount of activity in zone *z*, and InertCons reducing this effect when the previous quantity is small, other zonal attributes $\sum_{v \subset V} a_{a,v} \cdot X_{v,z}$), with *v* is the index of the other zonal attributes, *V* is the set of all other zonal attributes, and $X_{v,z}$ is one of the other zonal attributes, and the composite utility of technology options ($\alpha_{tech,a} \cdot \text{CUTech}_{a,z}$), with $\alpha_{tech,a}$ representing the sensitivity to the technology choices for activity a in zone *z*. The quantity of activity *a* in zone $z(W_{a,z})$ is then determined by the total activity $T_{a\zeta}$ allocated to the zones.

 Middle level: The middle level distributes activity quantities within each regional zone based on the technology options that determine the production and consumption rates of commodities. Each technology option represents a distinct production and consumption rate for different goods (Equations (3) and (4)).

$$\operatorname{Tech}_{p,a,z} = W_{a,z} \cdot \left(\frac{\exp(\lambda_{pa} \cdot \operatorname{UTech}_{p,a,z})}{\sum_{p \in p_a} \exp(\lambda_{p,a} \cdot \operatorname{UTech}_{p,a,z})} \right)$$
(3)

$$\text{UTech}_{p,a,z} = \frac{1}{\lambda_{p,a}} \cdot \ln\left(\text{OpSize}_{p,a}\right) + \text{UProd}_{p,a,z} + \text{UCons}_{p,a,z} \tag{4}$$

where Tech_{*p,a,z*} is the quantity of activity *a* in zone *z* applying technology option *p*; $W_{a,z}$ is the total activity *a* in zone *z*; $\lambda_{p,a}$ is the utility function dispersion parameter; UTech_{*p,a,z*} is the technology utility of activity a in zone *z* applying technology option *p*; OpSize_{*p,a*} is the relative size of technology option *p* for activity *a*, and UProd_{*p,a,z*} and UCons_{*p,a,z*} are the utilities from the production and consumption of commodities, respectively.

 Lowest Level: The lowest level allocates commodities among exchange locations where they are bought and sold. This involves two logit allocations per commodity in each zone, one for selling and one for buying. The exchange locations are influenced by prices and transportation impedance (Equations (5) and (6)).

$$S_{c,z,k} = TP_{c,z} \cdot \left(\exp(\lambda_s^c \cdot SU_{c,z,k}) / \sum_k \exp(\lambda_s^c \cdot SU_{c,z,k}) \right)$$
(5)

$$B_{c,z,k} = TC_{c,z} \cdot \left(\exp(\lambda_b^c \cdot BU_{c,z,k}) / \sum_k \exp(\lambda_s^c \cdot BU_{c,z,k}) \right)$$
(6)

where $S_{c,z,k}$ is the quantity of commodity *c* produced in zone *z* and allocated for sale at exchange location *k*, based on total production $TP_{c,z}$ and selling utility $SU_{c,z,k}$, with normalization across all exchange locations. Conversely, $B_{c,z,k}$ represents the quantity of commodity *c* consumed in zone *z* and allocated for purchase from exchange location *k*, using total consumption $TC_{c,z}$ and the buying utility $BU_{c,z,k}$, also normalized across exchange locations. λ_s^c and λ_b^c are the utility function dispersion parameters for selling and buying commodity c, respectively.

The PECAS AA model uses several key input parameters to adjust and improve the performance of the model. These parameters include dispersion parameters for buying (λ_b^c) and selling (λ_s^c) , from the nested logit model, constants of the discrete choice model, and option weights that adjust the sizes of technology options (OpSize_{*p,a*}) within the nested logit model [60,67]. Adjustments to these parameters are made iteratively to ensure that the model closely aligns with base-year production data as accurately as possible [60]. In the PECAS AA model, exchange zones represent markets where total supply—comprising both selling allocations and imports, which respond to fluctuations in exchange prices—meets total demand, consisting of buying allocations and exports. The algorithm works to find an equilibrium by adjusting the exchange prices within each zone until all markets reach equilibrium (Equations (7) and (8)).

The equilibrium solution is sought through a series of iterations, with this study considering a total of 3000 iterations. During each iteration, allocations are determined based on the current prices, which are subsequently updated according to the resulting supplies and demands. The algorithm calculates the partial derivatives of the total excess demand—defined as the surplus of demand by buyers and exporters over supply from sellers and importers—in each exchange zone with respect to the price in that specific zone. The derivatives with respect to prices in other zones or for other commodities are assumed to be zero, and the price change is then calculated [65].

To enhance the convergence, a step adjustment factor is used. If a price change reduces the total sum of the squared excess demand, the adjustment factor is slightly increased for the next iteration. Conversely, if the price change leads to an increase in excess demand, the adjustment is discarded, the factor is significantly reduced, and a smaller adjustment is applied in its place. This iterative process helps refine the equilibrium and improve the accuracy of the model's predictions [60,65].

$$TE_{c,k} = TDem_{c,k} = TSup_{c,k}$$
(7)

- $TE_{c,k}$ = exchange quantity for commodity c in exchange zone k;
- TDem_{c,k}= aggregate demand for commodity c in exchange zone k for all activities in the model area;
- TSup_{c,k} = aggregate supply of commodity c in exchange zone k for all activities in the model area.

$$\operatorname{Residual}_{c,k} = \operatorname{TSup}_{c,k} - \operatorname{TDem}_{c,k} \tag{8}$$

Residual_{c,k} = residual surplus quantity of supply over demand for commodity c in exchange zone k.

In theoretical equilibrium, the residual surplus for all commodities and exchange zones should be zero. In practice, however, to manage computational efficiency and avoid excessively long runtimes, the iterative process is terminated once the residual values meet the specified convergence criteria. These criteria indicate that the residuals are sufficiently close to zero.

Once this iterative process is completed, the PECAS AA model generates commodities output in the form of Open Matrix (OMX) matrices. These matrices are linked with Cube Voyager software. The monetary flows of the OD matrices are then converted into tonnage. This tonnage of OD matrices is subsequently loaded onto the multimodal transport super network, where it is used for mode and route choice analysis [67].



Figure 2. Jiangxi Province traffic analysis zones.

4.3. International (Import and Export) Commodity OD

Freight modeling has traditionally focused on urban areas or single province analyses, particularly in multimodal transportation contexts [70,71]. These studies, however, often neglect the influence of external regions, such as adjacent provinces and international trade corridors, on overall freight demand [25,72]. Therefore, this study expands its scope beyond Jiangxi's intra-provincial freight demand to consider the broader impacts on interregional and international freight transportation, including major import and export nodes such as seaports [73,74]. To estimate international import and export commodity OD matrices, we divided China into three regions [8,23,75].

1. First Region: Adjacent Provinces (First-Level External TAZs)

The first region consists of 13 provinces adjacent to Jiangxi, including Hubei, Hunan, Anhui, Zhejiang, Shanghai, Jiangsu, Chongqing, Sichuan, Guangdong, Fujian, Guangxi, Guizhou, and Henan (Table 1 and Figure 3) [68,74]. These provinces were selected due to their direct geographical proximity and their significant trade interactions with Jiangxi. Additionally, their inclusion aligns with the layout of key national railways and waterways, which are essential for medium- and long-distance freight movements. The underlying assumption is that freight follows the most efficient transportation routes, typically via railways and waterways, reflecting the actual interprovincial logistics patterns documented in previous studies [23,76].

2. Second Region: Outer Combined Provinces

The second region consists of the remaining provinces of China, which are grouped into seven broader clusters based on historical trade data from China's interregional IO tables and major transportation routes [69,76] (Table 1 and Figure 3). These groups are referred to as outer combined provinces and include Shandong, Hebei (which encompasses Beijing, Tianjin, Hebei, Shanxi, Liaoning, Jilin, and Heilongjiang); Shaanxi (Inner Mongolia, Shaanxi, and Ningxia), and Gansu, Yunnan, Tibet, and Hainan Provinces. The rationale for grouping these provinces is grounded in their historical data and strong economic and transportation linkages to Jiangxi and other provinces. This strategy simplifies the freight OD estimation by reducing the number of individual provinces, ensuring a manageable input while preserving the integrity of major trade corridors.

3. Third Region: International Import and Export Nodes

The third region consists of 21 key international seaports, including major export and import nodes along China's southeastern coast, such as Shanghai, Shenzhen, and Ningbo [77,78] (Table 1 and Figure 3). These seaports were selected based on their historical and current importance to international trade, as documented in sources such as the China Transportation Statistical Yearbook and the China Customs Statistical Yearbook [77–79]. These ports are crucial for understanding the interaction between domestic freight movement and international trade, particularly in terms of how goods flow among Jiangxi, other provinces, and global markets.

This categorization aimed to streamline the analysis by focusing on the eight major industries that are crucial to the agricultural and industrial sectors, thereby reducing the number of provinces from 31 to 21. The combination of import and export freight volumes was calculated using Equations (9) and (10), respectively.

$$IMC_{s,k} = \sum_{p \in s} IM_{p,k} \tag{9}$$

$$EXC_{s,k} = \sum_{p \in s} EX_{p,k} \tag{10}$$

where:

- *s* is one of the first set of combined provinces around a Jiangxi Province;
- k is the type of commodity;
- *p* is one of the second-set of provinces other than Jiangxi Province;
- $p \in s$ refers to the optimal path from Jiangxi Province to Province P through other provinces;
- IMC is the import freight volume of Jiangxi province after aggregations;
- *IM* is the volume of imported freight from other provinces to the seven combined provinces in the interregional IO table;
- EXC is the export freight volume of a Jiangxi Province after aggregations;
- EX represents the export freight volume from Jiangxi Province to another province in the interregional IO table.

From the provincial IO tables, we select commodities with inflows and outflows. However, the inflow/outflow and import/export data in the provincial IO table for each province consists of gross values [20]. Therefore, we disaggregated the domestic inflow and outflow using the interregional IO table for the total outputs of agriculture and industry. To achieve consistency with the provincial IO table, the interregional IO table for agriculture and industry was further subdivided into 12 commodities: agriculture (cereal, wood, and other agricultural products) and industry (coal and its products, metallic minerals, nonmetallic minerals (salt), oil, gas and its products, cement, chemical fertilizers and pesticides, steel and non-ferrous metals, mineral building materials, and other industrial products). This disaggregation was accomplished by using the ratio of the corresponding activities in each province, as shown in Equations (11) and (12). From the interregional IO table, the ratio of imports and exports for the specified commodity among provinces is calculated as the value of the commodity imported or exported from province i to province j, divided by the total value of the corresponding commodity [8,23].

Through this process, the flow of commodities from a specific province to the first set of adjacent provinces could be determined.

$$IM_{ii}^q = \alpha_i \times IM_{ii}^k \tag{11}$$

$$EX_{ij}^k = \alpha_i \times EX_{ij}^k \tag{12}$$

where:

- IM_{ij}^k is the import flow of commodity q from province i to j after splitting the interregional gross value of commodity k (q = (1,2,3 for agriculture and 4–12 for industry), (k = agriculture and industry).
- EX_{ij}^k is the export flow of commodity q from province i to j after splitting the interregional gross value of commodity k (q = (1,2,3 for agriculture and 4–12 for industry), (k = agriculture and industry).
- α_{i} , and α_{i} are the inflow and outflow ratios of the corresponding province from MRIO.
- *IM* and *EX* are the total of IO table import and export of different commodity k.

To model freight flows between provinces and international markets, including seaport entry and exit points, we employed a gravity model [46]. Initially, an AEF table was developed using an interregional IO table to quantify commodity values (in 10,000 yuan/year). These values were then converted into freight volumes (tons/year) using conversion factors from previous studies [67]. This study focuses on 21 major southeast coastal ports, identified from the China Transportation Statistical Yearbook, which play a critical role in the trade network (see Table 1 and Figure 3). Given the prominence of railway and maritime routes for international trade, these ports serve as primary hubs for imports and exports [80]. Furthermore, an external zone was established for each port to facilitate the management of its interactions with foreign imports and exports, integrating the connectivity of each port to this zone and the allocation of imports and exports. An integrated transportation supernetwork was developed, incorporating physical networks-highway, railway, and waterway—as well as virtual links and nodes, including connectors, transfer links, and connections between these transportation networks and the TAZ centroids. The supernetwork combines transportation skims, 2012 port throughput data, 2012 IO tables, 2012 AEF table, and conversion factors to compute integrated transportation impedances.

The gravity model estimates OD matrices for trade flows between provinces and international ports, based on the assumption that trade is proportional to production at the origin, consumption at the destination, and the generalized cost of transportation or other forms of impedance between regions [2,18]. Due to differences in routes, costs, and travel times across various transportation modes, generalized costs are applied to adjust the weights accordingly. The generalized cost function incorporates parameters such as transportation distances, times, and costs across three key modes: highway, railway, and waterway [67,69]. These parameters are derived from an integrated multimodal transportation supernetwork developed using the Cube Voyager software, which captures the specific characteristics of each mode and commodity type [69]. The gravity model formulation and generalized cost function are presented in Equations (13) and (14), respectively.

$$q_{ij} = O_i \cdot \frac{D_j \cdot f(c_{ij})}{\sum_{j=1}^n D_j \cdot f(c_{ij})}$$
(13)

$$f(c_{ij}) = \sum_{m=1}^{3} (Dis_{ij}^m \times KD^{c,m} + Time_{ij}^m \times KT^{c,m})$$
(14)

where:

qij is freight flow between province i and port j;

- $f(c_{ij})$ is the generalized transportation cost between the province i and port j, considering the mode-specific distance Dis_{ij}^m and time $Time_{ij}^m$;
- $KD^{c,m}$ and $KT^{c,m}$ are distance and time cost for commodity c under the mode m;
- *Oi* and *Dj* are the freight production and attraction volumes, respectively;
- Dis_{ii}^m is the distance between the port i and province j under the mode m;
- *n* is number of the TAZs.



Figure 3. Multimodal transportation network in Jiangxi intra-regional, and international ports.

Table 1. Jiangxi Provir	ice and combinatio	ons of other pro	ovinces and	ports
-------------------------	--------------------	------------------	-------------	-------

S. No	Jiangxi Region and Its Adjacent 13 Provinces	Combinations of Provinces	International (Import and Export) Ports
1	Shanghai	Shanghai	Shanghai
2	Jiangsu	Jiangsu	Lianyungang
3	Zhejiang	Zhejiang	Jiaxing, Ningbo–Zhoushan, Taizhou, Wenzhou
4	Anhui	Anhui	

S. No	Jiangxi Region and Its Adjacent 13 Provinces	Combinations of Provinces	International (Import and Export) Ports
5	Fujian	Fujian	Fuzhou, Putian, Quanzhou, Xiamen
6	Henan	Henan	
7	Hubei	Hubei	
8	Hunan	Hunan	
9	Guangdong	Guangdong	Shantou, Huizhou, Shenzhen, Humen, Guangzhou, Zhongshan, Zhuhai, Jiangmen, Maoming, Zhanjiang
10	Guangxi	Guangxi	Beibu Bay Port
11	Chongqing	Chongqing	
12	Sichuan	Sichuan	
13	Guizhou	Guizhou	
14	Hebei	Beijing, Tianjin, Hebei, Shanxi, Liaoning, Jilin, and Heilongjiang	
15	Shaanxi	Inner Mongolia, Shaanxi, and Ningxia	
16	Gansu	Gansu, Qinghai, Ningxia	
17	Shandong	Shandong	
18	Yunnan	Yunnan	
19	Tibet	Tibet	
20	Hainan	Hainan	

Table 1. Cont.

4.4. Development of Multimodal Transportation Network

In this study, we developed a multimodal transportation network for Jiangxi Province by integrating highways, railways, and waterways. The planning area was divided into TAZs based on district and county boundaries using GIS spatial operations. The network was built using ESRI's ArcGIS 10.8 (www.esri.com, accessed on 23 September 2024) and Cube Voyager software (https://www.bentley.com/software/openpaths, accessed on 23 September 2024), with shapefiles representing different transportation modes. The network features 20 link function classes across highways, railways, and waterways, reflecting infrastructure types such as motorways, national highways, waterways, and railway lines. The highway network includes expressways, national highways, provincial roads, main roads, and secondary roads. The waterway network comprises navigable waterways, locks, and ports (Jiujiang, Nanchang, Ji'an, and Ganzhou). The railway network includes railway lines and stations. The TAZs are linked to the highway network through centroid connectors, while ports and railway stations are connected to their respective networks (waterways and railways) via virtual intermodal links. These virtual links are crucial in representing the multimodal interactions between freight modes, allowing seamless transitions between highways, railways, and waterways. Furthermore, virtual transfer links are used to simulate multimodal transfers, creating an integrated supernetwork that supports the efficient modeling of multimodal freight flows [81]. Figure 4 shows the structure of the Jiangxi multimodal transportation network.



Figure 4. Multimodal transportation network in Jiangxi province.

Development of Multimodal Incremental Freight Assignment Model

This study developed a multimodal incremental freight assignment method for highways, railways, and waterways following the methodology outlined by Yamaguchi et al. [53] (Figure 5). Incremental freight assignment assigns traffic volumes across multiple steps. In each step, a fixed proportion of the total demand is assigned to the network using an all-or-nothing approach. After each step, the link travel times are recalculated based on the updated traffic volumes, ensuring that the congestion effects are captured progressively. This method iteratively allocates freight flows to the optimal paths by minimizing composite costs, which include loading costs, transfer fees, and an alternative specific constant (ASC) term (Equation (15)). A multimodal network incorporates different cost functions for highways, railways, and waterways, each with mode-specific impedance functions. For highways, the Bureau of Public Roads (BPR) function models the relationship between the road travel time and traffic flow [82], as shown in Equation (16). For railways, a modified Davidson function accounts for the capacity constraints and congestion effects on rail travel times (Equation (17)). For waterways, travel times are calculated based on path length, vessel speed, and additional factors influencing waterborne transport (Equation (18)). The ASC term is used to aggregate the link-specific values for each mode, ensuring that all relevant costs are captured. The freight flows are assigned across the super network, respecting capacity constraints, and link travel times are recalculated iteratively until all demands are fully allocated. The final freight assignment considers the interaction between different modes, the costs of mode transfers, and the congestion effects on each network link. To evaluate the performance of the model, we employed standard goodness-of-fit metrics, including the coefficient of determination (R²) and relative error (RE), as given in Equations (19) and (20), respectively.

$$C_{ij}^{(c,r)} = \sum_{(a,b)}^{d} \left(t_{ij}^{(c,r)} \times LF_{ab}^{c} \right) + \sum_{(k,l)}^{n} \left(t_{ij}^{(c,r)} \times L_{kl}^{(c,m)} \times F^{(c,m)} \right) + \sum_{(o,p)}^{q} \left(t_{ij}^{(c,r)} \times TF_{op}^{(c,FT)} \right) + t_{ij}^{(c,r)} \times VPT^{c} \times \frac{ROR_{t}}{365 \times 24} \times \left[\sum_{(a,b)}^{d} LT_{ab}^{c} + \sum_{(k,l)}^{n} T_{kl}^{(c,m)} + \sum_{(o,p)}^{q} TT_{op}^{(c,FT)} \right] + ASC$$
(15)

where;

- $C_{ij}^{(c,r)}$ denotes the cost function of commodity *c* from zone *i* to zone *j* via path *r*; note : a given OD pair may have multiple valid paths, so $r \in R$;
- $LF_{(ab)}^{(c)}$ is the cost per ton for loading commodity *c* on *transfer* arc(*a*, *b*);
- $F^{(c,m)}$ represents the cost per kilometer (RMB/km) for transporting c by mode m = highway, waterway, and railway;
- $t_{ij}^{(c,r)}$ is the weight of *c* transported via route *r* in path (i, j);
- $L_{(k,l)}^{(m)}$ is the distance of the road link (k, l) for mode m;
- $LT^{(c)}_{(a,b)}$ is the loading time (hours) on arc(a, b) in path (i, j);
- $T_{(k,l)}^{(c,m)}$ is transportation time (hours) spent in passing through the (k,l) arc. Note: this transportation time should be the link (k,l) obtained by the BPR or similar function used by each mode when the flow assigned to a path is $t_{i,j}^{c,r}$ through the network flow distribution (k,l) transportation time on the link;
- TF^(c,FT)_{op} is intermediate unit commodity transfer fee (RMB/ton) for transporting commodity c from zone i to zone j via route r, where the link (o,p) consisting of nodes o and p belongs to path r, and the road link belongs to the transit link;
- TT^(c,FT) is the intermediate transfer time (hours);
- VPT^c is the value of commodity c;
- ROR_t is the annualized rate of return;
- *ASC* is the alternative specific constant term.

$$\begin{cases} t_{s,l}^{c,H} = \frac{t_{s,l}^{c,H}(0)}{\alpha_1} \cdot \left[1 + \left(q_{s,l}^{c,H} / C_{s,l}^H \right)^{\beta} \right] \\ \beta = \alpha_2 + \alpha_3 \cdot \left(q_{s,l}^{c,H} / C_{s,l}^H \right)^3 \end{cases}$$
(16)

where:

— $t_{(s,l)}^{(c,H)}$ is the travel time (hours) for commodity *c* on highway (H) link (s, l).

— $t_{(s,l)}^{(c,H)}(0)$ is the free-flow travel time (hours) for commodity *c* on highway link (s,l).

- α_1 is a scaling factor;
- $q_{(s,l)}^{(c,H)}$ is the flow of commodity *c* on highway link (s, l).
- $C_{(s,l)}^{(H)}$ is the capacity of highway link (s, l)
- β determines the rate at which the travel time (hours) increases with traffic-flow approach capacity;

— α_2 and α_3 are additional parameters.

$$t_{s,l}^{c,R} = t_{s,l}^{c,R}(0) \cdot \left[1 + \xi \cdot \left(\frac{q_{s,l}^{c,R}}{C_{s,l}^{c,R} - q_{s,l}^{c,R}} \right) \right]$$
(17)

where:

- $t_{(s,l)}^{(c,R)}$ is the travel time (hours) for commodity *c* on railway (R) link (s,l);
- $t_{(s,l)}^{(c,R)}(0)$ is the base (free-flow) travel time (hours) for commodity *c* on the railway link (s,l);
- ξ is a parameter representing the impedance factor, which influences the rate at which travel time (hours) increases with congestion;

q_(s,l)^(c,R) is the flow (or demand) of commodity *c* on railway link (*s*, *l*) during period *R*;
 C_(s,l)^(c,R) is the capacity of railway link (*s*, *l*) for commodity *c*.

$$t_{e,g}^{c,W} = \sum_{f_y \in (e,g)} \frac{L_{f_y}^{c,W}}{V_{f_y}^{c,W}} + B_{e,g} \cdot t_0$$
(18)

where:

- $t_{(e,g)}^{(c,W)}$: Travel time (hours) for commodity *c* on waterway (w) link (e,g);
- $\sum_{f_y \in (e,g)}$: Summation over all paths f_y from node e to node g;

— $L_{(f_y)}^{(c,W)}$: Length of path f_y for commodity *c* in the waterway;

- $V_{(f_y)}^{(c,W)}$: Flow of commodity *c* on path f_y in the waterway;
- $B_{(e,g)}$: Bias term representing additional time not accounted for by the path length or flow.
- t_0 : Base (free-flow) travel time (hours) on the waterway link.

$$R^{2} = 1 - \frac{\sum_{i=1}^{n} (y_{i} - \dot{y}_{i})^{2}}{\sum_{i=1}^{n} (y_{i} - \dot{y}_{i})^{2}}$$
(19)

$$RE = \left| \frac{\dot{y}_i - y_i}{y_i} \right| \times 100$$
(20)

where;

- \dot{y}_i is the estimated value of the ith observation;
- y_i is the observed value for the i-th observation;
- \overline{y} is the mean of the observed values.



Figure 5. Schematic diagram of the incremental freight assignment method.

5. Results and Discussion

This study proposed an integrated framework for estimating the OD matrices of multimodal, multi-commodity freight flows in regional and international import/export contexts. The integrated framework was implemented and tested in a case study of Jiangxi Province.

5.1. Results of Regional and International Multi-Commodity OD Flows

This section presents an analysis of the OD flows of key commodities in the regional, interregional, and international contexts. Commodities analyzed included cereal, wood, agricultural products, coal and its products, metallic minerals, non-metallic minerals (salt), oil and gas and their products, cement, chemical fertilizers and pesticides, steel and non-ferrous metals, mineral building materials, and other industrial products. The flows were visualized using Python 3.8 and ArcGIS 10.8, providing critical insights into the spatial distribution of these commodities. The findings highlight the interconnectedness of the Jiangxi regional economy, as well as the role of major ports and provinces in facilitating both domestic and international trade.

1. Cereal

The OD analysis of cereal commodity domestic inflows/outflows and foreign imports/exports yields significant findings (Figure 6a). Shandong Province is the largest importer, receiving 64,068 tons/day of cereals from Hebei Province. Furthermore, Zhejiang imports 17,593 tons/day and Henan imports 17,138 tons/day, mainly from Hebei. Guangdong imports 8332 tons/day from Guangzhou Port and 7328 tons/day from Beibu Bay Port. Similarly, Jiangsu imports 5512 tons/day from Guangzhou Port and 4812 tons/day from Beibu Bay Port. Conversely, Hebei is the primary exporter, dispatching 17,161 tons/day to Shaanxi Province. Additionally, Guangdong exports 82 tons/day to Beibu Bay Port, while Jiangxi exports 50 tons/day to Beibu Bay Port. The OD results indicate that Ports such as Guangzhou and Beibu Bay play pivotal roles, serving as gateways for cereal imports into Guangdong and Jiangsu. The dominance of Hebei as a central player in the cereal trade aligns with its strategic geographic position and extensive agricultural production capacity. This pattern underscores the region's critical role in balancing the cereal demand across different provinces, while ports in southern China, such as Guangzhou, strengthen international trade linkages [83,84].

2. Wood

The OD analysis of wood commodities revealed significant findings regarding domestic inflows/outflows and foreign imports/exports (Figure A1a, Appendix A). Shandong Province is the leading importer, receiving 25,830 tons/day from Hebei Province, followed by Henan with 6909 tons/day and Guangdong with 6326 tons/day from Hebei. Guangdong imports 3976 tons/day from Beibu Bay Port and 3345 tons/day from Xiamen Port. Shanghai imports 2270 tons/day from Shanghai Port. Hebei Province is a major exporter, sending 6819 tons/day to Shaanxi Province. Fujian exports 4169 tons/day to Lianyungang Port, while Guangdong exports 1606 tons/day to the same port. Guangxi exports 431 tons/day to the Lianyungang Port. The prominence of Hebei as a major supplier reflects its extensive forestry resources, while coastal provinces like Guangdong and Fujian capitalize on port facilities to support both domestic and international wood trade. These patterns align with the existing literature that emphasizes the importance of Hebei and Guangdong in facilitating wood distribution, leveraging their strategic locations and port infrastructure [85].

3. Other Agricultural Products

The OD analysis of other agricultural commodities revealed significant findings on domestic inflows/outflows and foreign imports/exports (Figure A1b, Appendix A). Shandong Province is the largest importer, receiving 73,276 tons/day from Hebei Province. Shaanxi and Henan Provinces import 19,627 tons/day and 19,601 tons/day, respectively, from Hebei. Zhejiang Province imports 18,012 tons/day from Hebei Province. Shandong imports 16,129 tons/day from Shaanxi Province. Guangdong Province imports 2469 tons/day from Beibu Bay Port and 2840 tons/day from Guangzhou Port. On the export side, Fujian Province exports 740 tons/day to Beibu Bay port, followed by Guangdong at 299 tons/day and Hubei at 180 tons/day. These findings highlight Hebei's importance in domestic agricultural distribution and the strategic role of Guangdong's ports in facilitating international trade. This analysis underscores the interconnectedness of China's agricultural supply chain, with Hebei as a central supplier, and coastal provinces like Guangdong ensuring the flow of both imports and exports [83,86].

4. Coal and its Products

The OD analysis of coal and its products, domestic inflows/outflows, and foreign imports/exports revealed significant patterns (Figure 6b). Hebei Province is the leading importer, receiving 267,482 tons/day from Shaanxi Province. Guangdong imports 52,830 tons/day through Beibu Bay Port, 27,616 tons/day through Guangzhou Port, and 22,937 tons/day through Ningbo Zhoushan Port. Shanghai and Fujian are also importers, with Shanghai importing 22,709 tons/day through its port, Fujian importing 19,923 tons/day from Fuzhou Port, and 17,846 tons/day from Humen Port. Guangxi imports 15,137 tons/day through Beibu Bay Port. Shaanxi Province is a major exporter, sending 254,416 tons/day to Jiangsu Province. Hebei exports 205,604 tons/day to Jiangsu and 118,225 tons/day to Shandong. Shaanxi exports 150,922 tons/day to Shandong and 138,224 tons/day to Zhejiang, while Hebei exports 111,182 tons/day to Zhejiang. Fujian exports 946 tons/day to Shanghai and 462 tons/day to Lianyungang, while Jiangsu exports 209 tons/day to Shanghai. These findings align with a previous study that identified thermal power plants, smelters, ceramic businesses, and non-ferrous metal companies as the major coal consumers in Guangdong Province [68]. Guangzhou Port in Guangdong Province, China's main coal import hub in 2021, highlights its logistical advantage of being almost equidistant between Inner Mongolia's largest coal mines and Indonesia [87].

5. Oil, Gas, and their Products

The OD analysis of domestic inflows/outflows, foreign imports, and exports of oil, gas, and their products yields interesting findings (Figure A1c, Appendix A). Guangdong Province imports 249,367 tons/day from Ningbo Zhoushan Port, 72,147 tons/day from Huizhou Port, 54,150 tons/day from Zhanjiang Port, and 42,074 tons/day from Maoming Port. Hebei Province exports 147,792 tons/day to Shandong, 72,225 tons/day to Jiangsu, and 54,514 tons/day to Zhejiang. Zhejiang exports 11,010 tons/day to Ningbo Zhoushan Port, while Hebei sends 5,140 tons/day to Gansu. Fujian Province exports 3160 tons/day to Zhuhai port. The findings highlight the interconnectedness of China's energy supply chain, with Guangdong and Hebei acting as crucial nodes for both imports and exports of oil and gas [88,89]. The strategic role of coastal ports like Ningbo Zhoushan is vital for managing the flow of these critical energy resources.

6. Metallic Minerals

The OD analysis of domestic inflows/outflows and foreign imports/exports of metallic mineral commodities yields intriguing findings (Figure 6c). Shanghai imports 94,069 tons/day through the Port of Shanghai. Hebei Province imported 87,430 tons/day from Guangdong Province, underscoring the importance of its supply. Zhejiang Province and Shanghai import 65,753 and 57,276 tons/day, respectively, via Ningbo Zhoushan port, followed by Jiangsu Province with 53,461 tons/day. Hebei Province is a major exporter, sending 265,765 tons/day to Shandong Province, and 211,498 tons/day to Jiangsu Province. Guangdong Province exports 116,793 tons/day to Guangxi Province. Anhui and Hubei Provinces export 45,127 and 41,618 tons/day, respectively, to Jiangsu Province. This analysis highlights the strategic roles of Guangdong in both importing and exporting metallic minerals, with key ports such as Shanghai and Ningbo Zhoushan acting as crucial gateways for these commodities. The trade flows indicate a highly integrated supply chain that is essential for sustaining industrial production across multiple provinces. This flow pattern is consistent with a previous study that highlight the shift of metallic mineral transportation from overland routes to waterway routes, optimizing the use of key ports along the Yangtze River and coastal regions [90].

7. Non-metallic Minerals

The OD analysis of domestic inflows/outflows of non-metallic minerals and foreign imports/exports revealed significant findings (Figure A1d, Appendix A). Jiangsu Province imports 197,547 tons/day from Hebei Province, indicating a high demand. Hebei Province exports 49,062 tons/day to Shaanxi Province, 27,950 tons/day to Shandong Province, and 25,028 tons/day to Zhejiang Province, underscoring its role as a major supplier. Gansu Province exports 22,751 tons/day to Jiangsu Province, highlighting robust interprovincial trade flows. Fujian Province exports 3159 tons/day, and Guangdong Province exports 2089 tons/day to Beibu Bay Port, contributing to international trade dynamics. This analysis indicate the importance of Hebei as a major exporter and Jiangsu as a critical consumer of non-metallic minerals, while coastal provinces like Fujian and Guangdong maintain essential links to international trade routes [90,91].

8. Chemical Fertilizers and Pesticides

The OD analysis of domestic inflows/outflows and foreign imports/exports of chemical fertilizers and pesticides revealed significant results (Figure A1e, Appendix A). Anhui Province imports 3726 tons/day from Jiangsu Province. Jiangsu exports 7388 tons/day to Hebei Province, while Guangdong exports 6520 tons/day to Guangxi Province. Hebei exports 5078 tons/day to Jiangsu and 3156 tons/day to Chongqing. Jiangsu exports 6199 tons/day to Yunnan and 3710 tons/day to Beibu Bay port in Guangdong. This analysis highlights the strategic roles of Jiangsu, Hebei, and Guangdong in facilitating both regional and international flows of chemical fertilizers and pesticides, reflecting a well-integrated supply chain that is crucial for agricultural productivity across China [92,93].

9. Cement

The OD analysis of domestic inflows/outflows and foreign imports/exports of cement revealed significant results (Figure 6d). Zhejiang Province imports 56,335 tons/day of these products from Henan Province. On the export side, Henan Province was a significant contributor, exporting 117,522 tons/day of cement to Jiangsu Province and 49,388 tons/day to Yunnan Province. Hebei Province is also a major exporter, sending 115,271 tons/day of cement to Jiangsu Province, 62,923 tons/day to Zhejiang Province, and 44,455 tons/day to Chongqing. Additionally, Guangdong Province exports 937 tons/day, and Jiangsu Province exports 891 tons/day to Beibu Bay Port. This analysis demonstrates the central role of Henan and Hebei as major cement suppliers, with robust trade flows to key industrial provinces like Jiangsu and Zhejiang, while coastal regions like Guangdong and Jiangsu maintain international trade connections [94].

10. Mineral Building Materials

The OD analysis of domestic inflows/outflows and foreign imports/exports of mineral building materials revealed these results (Figure A1f, Appendix A). Guangdong province imports 2310 tons/day, and Fujian province imports 1884 tons/day, both from Xiamen port. On the export side, Gansu province is a significant sender, exporting 35,878 tons/day of mineral building materials to Shanghai, 29,297 tons/day to Zhejiang province, 22,025 tons/day to Jiangsu province, 19,535 tons/day to Shaanxi province, and 16,291 tons/day to Chongqing. Additionally, Guangdong Province plays a notable role in exporting 11,703 tons/day of mineral building materials to the port of Xiamen. Fujian Province exports 8155 tons/day to Fuzhou Port and 7913 tons/day to Xiamen Port. This analysis highlights the significant role of inland provinces like Gansu in supplying mineral building materials to international markets through their ports [95].

11. Steel and Non-Ferrous Metals

The OD analysis of domestic inflows/outflows, as well as foreign imports/exports of steel and non-ferrous metals, revealed intriguing results (Figure 6e). Guangdong province imports 126,271 tons/day from Ningbo Zhoushan Port and 75,551 tons/day from Lianyun-gang Port. Shanghai imports 82,386 tons/day from its port and 70,574 tons/day from Ningbo Zhoushan port, underlining the importance of these coastal hubs in supplying steel and non-ferrous metals to key industrial regions. Hebei Province exports 114,169 tons/day to Jiangsu Province, and 68,462 tons/day to Zhejiang Province. Guangdong exports 168 tons/day to Zhanjiang port, Jiangsu exports 111 tons/day to the same destination, and Shanghai exports 106 tons/day to its port (Shanghai port). This analysis emphasizes the strategic role of coastal ports like Ningbo Zhoushan and Lianyungang in supporting heavy industrial demand in Guangdong and Shanghai, while Hebei continues to be a key exporter of steel and non-ferrous metals to neighboring provinces.

12. Other Industrial Products

The OD analysis of domestic inflows/outflows and foreign imports/exports of other industrial products reveals intriguing results (Figure 6f). Shanghai is a significant destination, importing 216,778 tons/day of other industrial products through the Shanghai port. Guangdong imports 84,315 tons/day from Shenzhen Port and 60,992 tons/day from Ningbo Zhoushan Port. On the export side, Shanghai exports 188,323 tons/day to its port (Shanghai Port). Hebei exports 74,561 tons/day to Jiangsu, 71,483 tons/day to Shanghai, and 72,978 tons/day to Guangdong. Guangdong exports 171,513 tons/day to Guangxi, 120,402 tons/day to Shenzhen Port, and 85,811 tons/day to Ningbo Zhoushan Port. Jiangsu

exports 78,372 tons/day to Shanghai, while Guangdong exports 67,718 tons/day to Shanghai. This analysis indicates the central role of Shanghai in both importing and exporting industrial products, along with significant export activities from Guangdong and Hebei. The flow patterns indicate Shanghai's dominance as a trade hub, with notable contributions from Guangdong and Hebei to the broader trade network.

Figure 6. Commodity OD flows of imports and exports among provinces and international ports: (a) cereal; (b) coal products; (c) metallic minerals; (d) cement; (e) steel and non-ferrous metals; (f) other industrial products.

5.2. Results of the Multimodal Incremental Freight Assignment Model

The estimated OD flows are allocated to the multimodal transportation network. This assignment process integrates the mode-split and freight distribution, ensuring an accurate representation of freight flows within the model. Freight assignment involves allocating freight volumes between each OD pair within a multimodal transportation supernetwork consisting of highways, railways, and waterways. This study utilized Cube software and its GIS-supported select link analysis tool to identify the relationship between the assigned traffic on specific network links and the corresponding OD-specific freight flows routed over those links [6]. This allocation provides detailed information on transport routes, transfer processes, and freight volumes of specific links [67]. The incremental assignment method used in this study incrementally assigns freight volumes to the network, which mitigates potential congestion that could arise from simultaneously assigning the entire OD volume at once. This method prevents unrealistic loading conditions and enhances the plausibility of the traffic assignment by progressively distributing flows. Consequently, the model avoids congestion-related inaccuracies and provides a more realistic simulation of freight movement across different network segments.

Model calibration followed approaches from relevant studies [67,69], adjusting for mode-specific shares across the transportation modes. To evaluate the accuracy of the assignment results, we calibrated the model by comparing the estimated mode share data to the observed traffic mode share in the Jiangxi Province. The comparison showed an error of less than 0.3% across different transport modes, demonstrating the model's high accuracy in representing real-world freight flows (Figure 7). This level of precision confirms the robustness of the incremental freight assignment method and its effectiveness in modeling multimodal transport systems.

Figure 7. Modal split of freight transportation for Jiangxi province.

The freight assignment results for the intra-regional (99TAZs), interregional (20 TAZs), and international (import and export 21 TAZs) flows are presented using the incremental freight assignment method (Figure 8a,b). The developed incremental multimodal freight assignment model systematically assigns freight traffic across highways, railways, and waterways, using a composite cost function. As described in Equation (15), the composite

cost function incorporates four primary attributes: loading costs, transfer costs, modalspecific constant terms, and value per ton. Loading fees, transfer fees, and value per ton are integrated into the model to accurately represent the costs incurred at different stages of transportation.

Figure 8. Freight assignment results: (a) Jiangxi intra-region; (b) international (import and export).

To account for the interaction between vehicular flow and travel time on highways and railways, the model employs an adapted BPR function for roadways (Equation (16)) and the Davidson impedance function for railways (Equation (17)), which considers the volume-to-capacity ratio of each segment. Additionally, a novel BPR function was introduced for waterway navigation (Equation (18)), which estimates travel time by considering the path length and commodity flow, with an additional bias term to account for unique waterborne transport conditions.

From the freight assignment results of the Jiangxi model, it is evident that the Beijing-Kowloon Railway and the Ji-Guangzhou Expressway serve as the primary freight channels for the north-south corridor in Jiangxi Province. Additionally, the Shanghai-Kunming Expressway acts as the main freight channel for the east–west corridor in the province. Furthermore, the freight assignment results from the extended model indicate that the Beijing–Guangzhou railway and sea transport are the primary freight channels in the north-south region. This observation is consistent with the actual traffic situations. Finally, the R^2 of the estimated freight volumes of the specific links by mode was calculated for further validation. To further validate the model, the estimated freight volumes on specific links by transport modes were compared with observed data, such as the Annual Average Daily Traffic (AADT) for highways, railway station data, and port throughput data. It can be seen from Figure 9a-c that the R² of estimated freight volumes on specific links for highways, railways, and waterways are 0.953, 0.925, and 0.931, respectively. This indicates that the high R^2 of freight volumes on specific links by mode suggests that the model performs well in predicting freight volumes. These high R² values affirm the model's reliability and highlight its potential for real-world freight planning and policy-making. By accurately identifying key freight corridors, such as the Beijing–Kowloon Railway and the Shanghai-Kunming Expressway, the model provides critical insights for infrastructure

investment and capacity planning. Enhancing the capacity of these primary channels can significantly improve freight mobility and reduce transportation costs, thereby ensuring a more efficient regional logistics network.

Figure 9. Comparative results of observed and estimated OD flows of freight: (**a**) highway; (**b**) waterway; (**c**) railway.

The use of a super-network approach, in which all transportation modes are integrated into a single network model, offers a significant advancement over traditional multimodal freight assignment models. This method enables a more detailed representation of intermodal operations, such as loading, unloading, and transshipment, through the use of virtual links [82]. By incorporating these logistical processes, the model delivers a more comprehensive assessment of freight flow dynamics, including the cost and time implications of the mode transfers. This approach is particularly useful for understanding complex freight movement patterns across multiple transportation modes, especially in a region like Jiangxi, where a diverse set of transport infrastructure—highways, railways, and waterways—plays a pivotal role [96]. Furthermore, the results emphasize the importance of improving intermodal connectivity, particularly at transfer points between the highway, railway, and waterway modes. These findings are crucial for developing strategies to enhance the resilience of Jiangxi's freight transportation system, particularly considering the growing freight demand driven by economic expansion and international trade. The integration of multiple transportation modes into a single modeling framework provides a holistic understanding of freight flow dynamics, which is essential for addressing the challenges of modern freight transportation.

Trip Length Distribution (TLD) calibration was performed for various transportation modes, including highways, waterways, and railways, as shown in Figure 10a,c. The observed TLD data were sourced from diverse large-scale datasets: truck GPS data for highways, waybill data for railways, and vessel tracking data for waterways [69,97,98]. For the truck GPS data, trip lengths were determined based on the recorded routes. Due to the absence of specific freight volume data, travel frequency was used as a weighting factor to establish the distribution pattern. The overall TLD for all commodity groups illustrates the

geographic distribution of economic activities such as production and consumption. Conversely, the TLDs for individual transportation modes reflect freight movement patterns across the extended region (Figure 10a,c). Notably, the shapes and ranges of TLDs vary significantly among transportation modes, thereby highlighting their distinct usage patterns.

Figure 10. Trip length distributions: (a) highway; (b) waterway; (c) railway.

Highway freight movements are predominantly intra-regional and regional, dominated by short-to-medium-distance trips, and relatively smooth TLDs. This pattern is likely influenced by factors such as cost, time, and convenience, suggesting that highway transportation is well suited for localized and varied movements (Figure 10a). This observation is consistent with findings of previous studies [9,99]. In contrast, the waterway freight movements exhibited a stable TLD pattern across a broad range of distances (Figure 10b). This stability reflects the nature of waterway transportation, where longer trips are typical and trip frequency remains consistent over extended distances. The fixed routes and infrastructure constraints, such as ports and designated waterways, contribute to this systematic pattern, making waterway transport less subject to local variations observed in highway transportation. Railway freight movements are primarily long-distance, particularly for bulk commodities, as evidenced by the mid-range spike in the observed TLD for rail transport (Figure 10c). However, the model may not fully capture the complexity of medium-distance rail movements, potentially due to competition with highways for shorter trips. The relative error between the observed and estimated TLDs was less than 15% for all three transportation modes. TLDs are influenced by specific land uses such as ports and agricultural areas, which create peaks in the data. These distributions reflect the distance ranges at which each transportation mode is most effective, with certain modes exceeding particular distances.

6. Conclusions

This study proposes a novel integrated framework for estimating freight flows in the intra-regional and international (import/export) contexts, linked through a two-tier multimodal transportation super network. This approach improves the accuracy of OD flow estimates by incorporating modeling techniques, and multisource data, including provincial and interregional IO tables, customs data, land use data, transportation networks, truck GPS data, railway waybills, ship visas, port statistical data, and statistical yearbooks. Consequently, it provides a more comprehensive estimation of OD matrices, particularly for long-distance transport of key commodities. The specific contributions of this study are threefold: First, a spatial economic model, PECAS activity allocation, was developed to estimate freight OD demand within a specific region. Second, the international (import and export) freight OD was estimated from different zones to foreign countries, including major import and export nodes, such as international seaports, using a gravity model with zone-pair friction obtained from a multimodal transportation model. Third, the OD matrices were converted from monetary value to tonnage and assigned to a multimodal transportation super network consisting of highways, railways, and waterways using the incremental freight assignment method.

The proposed framework was tested through a case study of Jiangxi Province, which is crucial for forecasting freight demand before the planning, design, and operation of the Ganyue Canal, acclaimed as a "project of the century". The framework was calibrated using traffic counts from highways, railways, and port-throughput data. The calibration results demonstrated high reliability, with an error of less than 0.3% for different transport modes in Jiangxi Province, indicating a high level of accuracy. To further verify the validity of the model, the estimated freight volumes on specific links by mode were compared with the observed data, including the AADT of highway links, railway station data, and port throughput data. The results showed R² values of 0.953 for highways, 0.925 for railways, and 0.931 for waterways, demonstrating the model's highly predictive performance for freight volumes on specific links. In addition, the calibration of the trip length distribution for extended regions was performed by integrating data from truck GPS, railway waybills, and ship visas. The relative error between the observed and estimated TLDs was less than 15% for all three transportation modes.

The findings of this study provide valuable insights that can guide and improve transportation infrastructure planning and policy-making in Jiangxi Province and similar regions. Prioritizing investments in multimodal transport networks, particularly by upgrading intermodal facilities and expanding critical freight corridors, is essential to meet growing freight demand. By enhancing key freight routes such as the Beijing-Kowloon Railway and the Shanghai–Kunming Expressway, policymakers can alleviate congestion, improve freight mobility, and reduce overall transportation costs. Additionally, promoting sustainable modal shifts, particularly towards railways and waterways for long-distance freight, can significantly mitigate environmental impacts. This shift is crucial for regions aiming to reduce greenhouse gas emissions and to align with national and international climate goals. To further optimize freight transport, policymakers should leverage the model's predictive capabilities to proactively address infrastructure bottlenecks and enhance data-driven planning. The model can forecast future freight demands, enabling targeted investments that align with regional economic needs. Moreover, aligning transport infrastructure development with strategic economic zones will boost regional trade and industrial growth, thus supporting economic expansion and resilience. All in all, these findings are crucial for policymakers and stakeholders involved in Jiangxi's long-term sustainable and efficient freight transportation infrastructure planning, underscoring the importance of proactive measures to accommodate future growth.

Within the proposed integrated framework, the PECAS AA model is based on a stochastic utility-based approach, enabling its application to any region with available data (IO tables, labor compensation, land use, employment data, transportation skims, and space data). However, a calibration process and adjustments may be necessary for regions with different economic structures or transportation networks. In regions where such data are scarce or less detailed, implementing the PECAS AA model may be challenging. Nevertheless, it is posited that a consistent treatment within a comprehensive framework is of even greater importance when data are not readily available.

The primary inputs for the PECAS AA model are derived from the AEF table. However, a significant challenge lies in systematically organizing and preparing the AEF table, which necessitates organizing and cleaning extensive datasets (economic, space, employment, and transportation). This study faced limitations with official government data, such as IO tables, employment, labor compensation, statistical yearbooks, and activity and commodity categories, which often exhibited inconsistencies and other shortcomings. A crucial takeaway was the importance of consistency checks to maintain data integrity, especially when aligning sectoral and commodity classifications in regional and interregional IO tables that vary across countries and regions. To further improve the generalizability of the model, region-specific parameters calibration is necessary. Specifically, utility dispersion parameters and technology option weights within PECAS AA should be adjusted to capture local economic and transport dynamics. Adjusting activity and commodity classifications, based on the regional economic structure, is also essential for capturing accurate intra-regional dynamics. Transportation network data can be obtained from open-source platforms such as OpenStreetMap, which provides a foundation for network development that can be extracted to the other regions. However, extensive data preprocessing is necessary, including the elimination of dangling links, realignment with the actual network, and proper alignment of network topology. Ensuring congruence between land use zones and TAZs is essential for accurate integrated modeling. In addition, we made simplifying assumptions to reconcile the provincial and interregional IO table data. In cases where such data are explicitly defined in other regions, these steps may be excluded, enabling the direct utilization of the data.

Furthermore, incorporating region-specific zone-pair frictions of multimodal transportation within the gravity model enables it to reflect the unique characteristics of local zones. International (imports and exports) OD flows, which consider a broader geographical scope, derive data from regional and interregional IO tables, port statistics, and customs data. However, challenges arise in developing countries such as China, where the nascent state of freight databases and integrated modeling impedes the study of economic exchanges between provinces. Obstacles such as data and comprehensive customs databases further complicate the process. Nevertheless, this study endeavored to provide a benchmark for import and export freight modeling, both regional and international, and presented results based on reliable multisource data. Ongoing advancements in information technology have facilitated the acquisition of the required data and utilization of state-of-the-art tools, which can enhance the accuracy and applicability of our models.

Overall, the proposed framework provides a more scientifically rigorous approach for the estimation of freight OD matrices in long-distance transport, albeit with significantly greater effort required for data preparation, model specification, calibration, and validation. This study focuses on Jiangxi Province and its international import and export nodes. Future research could extend this analysis to encompass the Yangtze River Economic Belt, a major trade corridor with significant regional, national, and international impacts. Additionally, the proposed framework can be applied to assess the effects of policy changes or infrastructure investments, such as the Ganyue Canal project, on freight demand.

Author Contributions: Conceptualization, M.S. and M.Z.; methodology, M.S., M.Z., Z.R., and J.D.H.; software, M.S. and Z.R.; validation, M.S., Z.R., and M.Z.; formal analysis, M.S.; investigation, M.S. and M.Z.; resources, M.S., Z.R., and M.Z.; data curation, M.S., Z.R., and M.Z.; writing—original draft, M.S. and M.Z.; writing—review and editing, M.S. and M.Z.; visualization, M.S. and Z.R.; supervision, M.Z. and J.D.H.; project administration, M.Z.; funding acquisition, M.Z. All authors have read and agreed to the published version of the manuscript.

Funding: This research was financially supported by the National Key R & D Program of China (grant no. 2021YFB2601300) and the National Natural Science Foundation of China (no. 52172309).

Data Availability Statement: The data supporting the findings of this study are openly available, as indicated by the URLs provided in the data section. Additionally, the data and related information may be obtained from the first author upon reasonable request.

125

125

125

Acknowledgments: The authors acknowledge the support of the Intelligent Transportation Systems Research Center, Wuhan University of Technology and Jiangxi Research Institute of Engineering and Technology Development Strategy for providing relevant data for our research.

Conflicts of Interest: The authors declare no conflicts of interest.

Appendix A

Figure A1. Commodity flows of imports and exports among provinces and international ports: (**a**) wood; (**b**) other agriculture products; (**c**) oil and gas products; (**d**) non-metallic minerals; (**e**) chemical fertilizers and pesticides; (**f**) mineral building materials.

References

- Pan, Y.; Darzi, A.; Yang, M.; Sun, Q.; Kabiri, A.; Zhao, G.; Xiong, C.; Zhang, L. National-Level Multimodal Origin–Destination Estimation Based on Passively Collected Location Data and Machine Learning Methods. *Transp. Res. Rec.* 2024, 2678, 525–541. [CrossRef]
- Beagan, D.F.; Tempesta, D.; Proussaloglou, K.; Systematics, C. *Quick Response Freight Methods*; No. FHWA-HOP-19-057; United States. Federal Highway Administration. Office of Operations: Washington, DC, USA, 2019. Available online: https://ops.fhwa. dot.gov/publications/fhwahop19057/fhwahop19057.pdf (accessed on 23 September 2024).
- Liao, Y.; Yeh, S.; Gil, J. Feasibility of estimating travel demand using geolocations of social media data. *Transportation* 2022, 49, 137–161. [CrossRef]
- 4. Levine, B.; Nozick, L.; Jones, D. Estimating an origin-destination table for US exports of waterborne containerised freight. *Marit. Econ. Logist.* **2009**, *11*, 137–155. [CrossRef]
- 5. Doustmohammadi, E.; Sisiopiku, V.; Anderson, M.D.; Doustmohammadi, M.; Sullivan, A. Comparison of freight demand forecasting models. *Int. J. Traffic Transp. Eng.* **2016**, *5*, 19–26.
- 6. Southworth, F. Freight flow modeling in the United States. *Appl. Spat. Anal. Policy* **2018**, *11*, 669–691. [CrossRef]
- Comi, A.; Coppola, P.; Nuzzolo, A. Freight Transport Modeling: Review and Future Challenges. In *Freight Transport Modeling*; Chapter 7; Springer: Berlin/Heidelberg, Germany, 2013; pp. 151–182.
- 8. Giuliano, G.; Gordon, P.; Pan, Q.; Park, J.; Wang, L. Estimating freight flows for metropolitan area highway networks using secondary data sources. *Netw. Spat. Econ.* 2010, *10*, 73–91. [CrossRef]
- 9. Holguin-Veras, J.; Thorson, E. Trip length distributions in commodity-based and trip-based freight demand modeling: Investigation of relationships. *Transp. Res. Rec.* **2000**, 1707, 37–48. [CrossRef]
- 10. Zhao, D.; Mihăiţă, A.-S.; Ou, Y.; Grzybowska, H.; Li, M. Origin–destination matrix estimation for public transport: A multi-modal weighted graph approach. *Transp. Res. Part C Emerg. Technol.* **2024**, *165*, 104694. [CrossRef]
- Liu, W.; Li, X.; Liu, H.; Tang, Z.; Guan, D. Estimating inter-regional trade flows in China: A sector-specific statistical model. J. Geogr. Sci. 2015, 25, 1247–1263. [CrossRef]
- 12. Lu, Q.-L.; Qurashi, M.; Antoniou, C. A two-stage stochastic programming approach for dynamic OD estimation using LBSN data. *Transp. Res. Part C Emerg. Technol.* 2024, 158, 104460. [CrossRef]
- 13. Veličković, M.; Stojanović, Đ.; Pantović, J. Modelling the urban freight flows for impact assessment of the urban consolidation centres by using the origin-destination matrices. *Transp. Res. Procedia* **2021**, *52*, 27–34. [CrossRef]
- 14. Sargento, A.L.M.; Ramos, P.N.; Hewings, G.J.D. Inter-regional trade flow estimation through non-survey models: An empirical assessment. *Econ. Syst. Res.* 2012, 24, 173–193. [CrossRef]
- Lawson, C.T.; Hu, P.S.; Parker, J.; Fan, C.-C.; Ford, C.; Linfors, B.; Conway, A.; Eisele, B.; Drumm, S.; Lee, J. 2020 Commodity Flow Survey Workshop [September 24, 2020]; National Research Council (US); Transportation Research Board: Washington, DC, USA, 2021; Available online: https://onlinepubs.trb.org/onlinepubs/circulars/ec269.pdf (accessed on 23 September 2024)National Research Council (US).
- Ramos, P.; Sargento, A. Estimating Trade Flows between Portuguese Regions Using an Input-Output Approach. In Proceedings of the 43rd Congress of the European Regional Science Association, Jyväskylä, Finland, 27–30 August 2003; pp. 27–30.
- 17. Lindall, S.A.; Olson, D.C.; Alward, G.S. Deriving multi-regional models using the IMPLAN national trade flows model. *J. Reg. Anal. Policy* **2006**, *36*, 76–83.
- Bachmann, C.; Kennedy, C.; Roorda, M.J. Estimating regional trade flows using commercial vehicle survey data. *Ann. Reg. Sci.* 2015, 54, 855–876. [CrossRef]
- 19. Park, J.; Gordon, P.; Moore, J.E.; Richardson, H.W. A two-step approach to estimating state-to-state commodity trade flows. *Ann. Reg. Sci.* **2009**, *43*, 1033–1072. [CrossRef]
- Zheng, H.; Bai, Y.; Wei, W.; Meng, J.; Zhang, Z.; Song, M.; Guan, D. Chinese provincial multi-regional input-output database for 2012, 2015, and 2017. *Sci. Data* 2021, *8*, 244. [CrossRef] [PubMed]
- 21. Yuan, Y.; Wang, Y.; Li, J.; Zhang, M. Input-Output Table and Input-output Model of Import and Export Internalization. *Adv. Manag. Appl. Econ.* **2023**, *13*, 297–319. [CrossRef]
- Bingham, P.; Maguire, A.; O'Rourke, L.; Pandey, B.; Markit, H.I.S. Freight Analysis Framework Commodity Flow Forecast Study (FAF Version 5): Final Forecasting Results; United States Department of Transportation, Federal Highway Administration: Washington, DC, USA, 2022. Available online: https://ops.fhwa.dot.gov/publications/fhwahop22037/fhwahop22037.pdf (accessed on 23 September 2024).
- Okamoto, N. In Non-Survey Method for Estimating a Multi-Regional Input-Output Model in China, 24th Applied Regional Science Conference, Japan, 2010; Supported by KAKENHI, Grant-in-Aid for Scientific Research (C) 21530233 and 22560538, Japan Society for the Promotion of Science (JSPS): Japan. 2010. Available online: https://www.iioa.org/conferences/20th/papers/files/ 895_20120125050_1201IIOAfullpaper.pdf (accessed on 15 July 2024).
- Fen, Z.; Dai, Z.-M. Correlation analysis of Jiangxi high technology industry and import and export trade. In Proceedings of the 2015 12th International Conference on Service Systems and Service Management (ICSSSM), Guangzhou, China, 22–24 June 2015; pp. 1–6.
- 25. Li, W.; Luo, C.; He, Y.; Wan, Y.; Du, H. Estimating Inter-Regional Freight Demand in China Based on the Input–Output Model. *Sustainability* **2023**, *15*, 9808. [CrossRef]

- 26. Ivanova, O.; Kancs, D.; Stelder, D. *Modelling Inter-Regional Trade Flows: Data and Methodological Issues in Rhomolo*; EERI Research Paper Series (No. 31/2009); Economics and Econometrics Research Institute (EERI): Brussels, Belgium, 2009. [CrossRef]
- 27. Vahidi, M.; Shafahi, Y. Time-dependent estimation of origin-destination matrices using partial path data and link counts. *Transportation* **2023**. [CrossRef]
- 28. Yang, F.; Jin, P.J.; Cheng, Y.; Zhang, J.; Ran, B. Origin-Destination Estimation for Non-Commuting Trips Using Location-Based Social Networking Data. *Int. J. Sustain. Transp.* **2015**, *9*, 551–564. [CrossRef]
- 29. Garrido, A.; Quintero-Espinosa, O.; Jaller, M. Obtaining the optimal origin-destination multimodal freight transportation network for the City of Bogotá. *Res. Transp. Bus. Manag.* **2023**, *49*, 101012. [CrossRef]
- 30. Zhang, C.; Wang, J.; Lai, J.; Yang, X.; Su, Y.; Dong, Z. Extracting origin—Destination with vehicle trajectory data and applying to coordinated ramp metering. *J. Adv. Transp.* 2019, 2019, 8469316. [CrossRef]
- Borgi, T.; Zoghlami, N.; Abed, M.; Naceur, M.S. Big data for operational efficiency of transport and logistics: A review. In Proceedings of the 2017 6th IEEE International Conference on Advanced Logistics and Transport (ICALT), Bali, Indonesia, 24–27 July 2017; pp. 113–120.
- 32. Holguín-Veras, J.; Patil, G.R. A multicommodity integrated freight origin–destination synthesis model. *Netw. Spat. Econ.* 2008, *8*, 309–326. [CrossRef]
- Kalahasthi, L.; Holguín-Veras, J.; Yushimito, W.F. A freight origin-destination synthesis model with mode choice. *Transp. Res. Part E Logist. Transp. Rev.* 2022, 157, 102595. [CrossRef]
- Al-Battaineh, O.; Kaysi, I.A. Commodity-based truck origin-destination matrix estimation using input-output data and genetic algorithms. *Transp. Res. Rec.* 2005, 1923, 37–45. [CrossRef]
- Ma, Y.; van Zuylen, H.J.; van Dalen, J. Freight Origin-Destination Matrix Estimation Based on Multiple Data Sources: Methodological Study (No. 12-0693). In Proceedings of the Transportation Research Board 91st Annual Meeting, Washington, DC, USA, 22–26 January 2012.
- 36. Teye, C.; Hensher, D.A. A commodity-based production and distribution road freight model with application to urban and regional New South Wales. *Transp. A Transp. Sci.* **2021**, *17*, 566–592. [CrossRef]
- Hensher, D.A.; Ho, C.Q.; Ellison, R.B. Simultaneous location of firms and jobs in a transport and land use model. *J. Transp. Geogr.* 2019, 75, 110–121. [CrossRef]
- Holguín-Veras, J.; Encarnación, T.; Ramírez-Ríos, D.; He, X.; Kalahasthi, L.; Pérez-Guzmán, S.; Sanchez-Díaz, I.; González-Calderón, C.A. A Multiclass Tour Flow Model and Its Role in Multiclass Freight Tour Synthesis. *Transp. Sci.* 2020, 54, 631–650.
 [CrossRef]
- Hunt, J.D.; Stefan, K.J. Tour-based microsimulation of urban commercial movements. *Transp. Res. Part B Methodol.* 2007, 41, 981–1013. [CrossRef]
- Wisetjindawat, W.; Sano, K.; Matsumoto, S.; Raothanachonkun, P. Micro-simulation model for modeling freight agents interactions in urban freight movement. In Proceedings of the 86th Annual Meeting of the Transportation Research Board, Washington, DC, USA, 21–25 January 2007; pp. 21–25.
- 41. Basso, F.; Pezoa, R.; Tapia, N.; Varas, M. Estimation of the Origin-Destination Matrix for Trucks That Use Highways: A Case Study in Chile. *Sustainability* 2022, 14, 2645. [CrossRef]
- 42. Tavasszy, L.A.; Stada, J.E. The Impact of Decreasing Border Barriers in Europe on Freight Transport by Road. In Proceedings of the 36th Annual Meeting of the Transportation Research Forum, Vols. 1 and 2, Daytona Beach, FL, USA, 3–5 November 1994.
- 43. Zhong, M.; Hunt, J.D.; Abraham, J.E. Design and development of a statewide land use transport model for Alberta. *J. Transp. Syst. Eng. Inf. Technol.* 2007, 7, 79–89. [CrossRef]
- Donnelly, R.; Upton, W.J.; Knudson, B. Oregon's Transportation and Land Use Model Integration Program. J. Transp. Land Use 2018, 11, 19–30. [CrossRef]
- 45. Zhong, M.; Hunt, J.D.; Abraham, J.E.; Wang, W.; Zhang, Y.; Wang, R. Advances in integrated land use transport modeling. In *Advances in Transport Policy and Planning*; Elsevier: Berlin/Heidelberg, Germany, 2022; Volume 9, pp. 201–230.
- Jadhav, S.; Ghosh, I. Future Prospects of the Gravity Model of Trade: A Bibliometric Review (1993–2021). Foreign Trade Rev. 2024, 59, 26–61. [CrossRef]
- 47. Masood, S.; Khurshid, N.; Haider, M.; Khurshid, J.; Khokhar, A.M. Trade potential of Pakistan with the South Asian Countries: A gravity model approach. *Asia Pac. Manag. Rev.* 2023, 28, 45–51. [CrossRef]
- 48. Nakano, S.; Nishimura, K. A nonsurvey multiregional input–output estimation allowing cross-hauling: Partitioning two regions into three or more parts. *Ann. Reg. Sci.* 2013, 50, 935–951. [CrossRef]
- Jackson, R.W.; Schwarm, W.R.; Okuyama, Y.; Islam, S. A method for constructing commodity by industry flow matrices. *Ann. Reg. Sci.* 2006, 40, 909–920. [CrossRef]
- 50. Osorio-Mora, A.; Núñez-Cerda, F.; Gatica, G.; Linfati, R. Multimodal capacitated hub location problems with multi-commodities: An application in freight transport. *J. Adv. Transp.* **2020**, 2020, 2431763. [CrossRef]
- 51. Uddin, M.; Huynh, N.N.; Ahmed, F. Assignment of Freight Traffic in a Large-Scale Intermodal Network under Uncertainty. *Highlights Sustain.* **2024**, *3*, 1–15. [CrossRef]
- 52. Thoen, S.; Tavasszy, L.; de Bok, M.; Correia, G.; van Duin, R. Descriptive modeling of freight tour formation: A shipment-based approach. *Transp. Res. Part E Logist. Transp. Rev.* 2020, 140, 101989. [CrossRef]

- 53. Yamaguchi, T.; Kawachi, K.; Shibuya, K.; Hagiwara, M.; Shibasaki, R. Global logistics intermodal network simulation modeling by incremental assignment and corridor development simulations in Myanmar. *Asian Transp. Stud.* **2023**, *9*, 100114. [CrossRef]
- Xia, X.-M.; Ma, X.-D. A Multimodal Multi-Product Transportation Network Model for Strategic Planning of Freight Flows. In Proceedings of the ICLEM 2012: Logistics for Sustained Economic Development—Technology and Management for Efficiency, Chengdu, China, 8–10 October 2012; American Society of Civil Engineers: Reston, VA, USA, 2012; pp. 197–202. [CrossRef]
- 55. Jain, A.; van der Heijden, R.; Marchau, V.; Bruckmann, D. Towards rail-road online exchange platforms in EU-freight transportation markets: An analysis of matching supply and demand in multimodal services. *Sustainability* **2020**, *12*, 10321. [CrossRef]
- 56. Pompigna, A.; Mauro, R. Input/Output models for freight transport demand: A macro approach to traffic analysis for a freight corridor. *Arch. Transp.* **2020**, *54*, 21–42. [CrossRef]
- 57. Singh, S.; Dwivedi, A.; Pratap, S. Sustainable Maritime Freight Transportation: Current Status and Future Directions. *Sustainability* **2023**, *15*, 6996. [CrossRef]
- 58. Venkadavarahan, M.; Raj, C.T.; Marisamynathan, S. Development of freight travel demand model with characteristics of vehicle tour activities. *Transp. Res. Interdiscip. Perspect.* 2020, *8*, 100241. [CrossRef]
- Crainic, T.G.; Damay, J.; Gendreau, M. An Integrated Freight Transportation Modeling Framework. In Proceedings of the International Network Optimization Conference 2007, Spa, Belgium, 22–25 April 2007; Available online: https://www.euroonline.org/enog/inoc2007/Papers/author.111/paper/paper.111.pdf (accessed on 15 July 2024).
- 60. Hunt, J.D.; Abraham, J.E. Design and application of the PECAS land use modelling system. In Proceedings of the 8th International Conference on Computers in Urban Planning and Urban Management (CUPUM), Sendai, Japan, 27–29 May 2003.
- 61. Echenique, M.H.; Flowerdew, A.D.J.; Hunt, J.D.; Mayo, T.R.; Skidmore, I.J.; Simmonds, D.C. The MEPLAN models of bilbao, leeds and dortmund. *Transp. Rev.* **1990**, *10*, 309–322. [CrossRef]
- 62. de La Barra, T.; Anez, J. The Mathematical and Algorithmic Structure of TRANUS. Unpublished Paper. 1998. Available online: http://www.modelistica.com (accessed on 10 July 2024).
- 63. Simmonds, D.C. The design of the DELTA land-use modelling package. Environ. Plan. B Plan. Des. 1999, 26, 665–684. [CrossRef]
- 64. Waddell, P. Urbansim: Modeling urban development for land use, transportation, and environmental planning. J. Am. Plan. Assoc. 2002, 68, 297–314. [CrossRef]
- 65. Hunt, J.D.; Abraham, A. PECAS—For Spatial Economic Modelling Theoretical Formulation System Documentation. 2009. Available online: https://www.hbaspecto.com/resources/PECAS-Software-User-Guide.pdf (accessed on 30 March 2024).
- 66. Tavasszy, L.A. Predicting the effects of logistics innovations on freight systems: Directions for research. *Transp. Policy* **2020**, *86*, A1–A6. [CrossRef]
- Ren, Z.; Zhong, M.; Cui, G.; Li, L.; Zhao, H. An Iterative Method for Calibrating Freight Value-per-Ton Conversion Factors of Integrated Land Use Transport Models Based on Multi-Source Data (No. TRBAM-24-03869). In Proceedings of the Transportation Research Board 103rd Annual Meeting, Washington, DC, USA, 7–11 January 2024.
- Liang, X.; Wang, S.; Sui, L.; Huang, Q. China's Ganyue Canal Construction: Forecast and Analysis of Its Transport Demand Based on Large Data. In Proceedings of the 3rd International Conference on Big Data and Informatization Education (ICBDIE 2022), Beijing, China, 8–10 April 2022; Atlantis Press: Amsterdam, The Netherlands, 2022; pp. 165–181.
- 69. Deng, G.; Zhong, M.; Raza, A.; Hunt, J.D.; Wang, Z. Design and Development of a Regional, Multi-commodity, Multimodal Freight Transport Model based on an Integrated Modeling Framework: Case Study of the Yangtze River Economic Belt. In Proceedings of the Transportation Research Board 101st Annual Meeting, Washington, DC, USA, 9–13 January 2022; Transportation Research Board: Washington, DC, USA, 2022.
- 70. Sharma, S.; Shelton, J.; Valdez, G.; Warner, J. Identifying optimal Truck freight management strategies through urban areas: Case study of major freight corridor near US-Mexico border. *Res. Transp. Bus. Manag.* **2020**, *37*, 100582. [CrossRef]
- SteadieSeifi, M.; Dellaert, N.P.; Nuijten, W.; Van Woensel, T.; Raoufi, R. Multimodal freight transportation planning: A literature review. *Eur. J. Oper. Res.* 2014, 233, 1–15. [CrossRef]
- 72. Shen, G.; Zhou, L.; Aydin, S.G. A multi-level spatial-temporal model for freight movement: The case of manufactured goods flows on the US highway networks. *J. Transp. Geogr.* **2020**, *88*, 102868. [CrossRef]
- Department, F.P. Jiangxi Daily Jiangxi Port Channel Improves Energy and Efficiency to Achieve "Big Through" [EB/OL]. Available online: https://swt.fujian.gov.cn/xxgk/jgzn/jgcs/dtgxdc/wszx/202210/t20221020_6019841.htm (accessed on 10 December 2022).
- 74. Ministry of Commerce (China). Doing Business in Jiangxi. 2023. Available online: http://english.mofcom.gov.cn/aroundchina/ jiangxi.shtml (accessed on 15 July 2024).
- 75. Egger, P.H.; Li, J.; Zhao, Y. Chinese regions' participation in global value chains and the associated global transmission of export price and quantity shocks. *Rev. Int. Econ.* **2024**, *32*, 371–393. [CrossRef]
- Pan, W.; Liu, Q. Spatial linkages of the Chinese economy. In Spatial Structure and Regional Development in China: An Interregional Input-Output Approach; Palgrave Macmillan: London, UK, 2005; pp. 101–127.
- World's Major Ports The Average Time of Ocean-Going International Container Ships in the World's Major Ports. 2022. Available online: https://new.qq.com/rain/a/20221012A043FV00 (accessed on 22 September 2022).
- 78. Jiang, M.; Zhao, S.; Jia, P. The spatial spillover effect of seaport capacity on export trade: Evidence from China pilot free trade zones. *Ocean. Coast. Manag.* 2023, 245, 106879. [CrossRef]

- 79. Xiao, R.; Liu, S.; Wu, L.; Luo, M.; Ma, R.; Li, J. Regional classification and competitiveness of port cluster: A case study of China's coastal ports. *Int. J. Logist. Res. Appl.* **2023**, 1–18. [CrossRef]
- 80. Yifan, Y. China-Europe Freight Train Service Bolsters Good Start of Chinese Economy in 2024; Belt and Road Portal: Beijing, China, 2024.
- Jourquin, B. A multi-flow multi-modal assignment procedure on large freight transportation networks. *Stud. Reg. Sci.* 2006, 35, 929–945. [CrossRef]
- 82. Jourquin, B.; Limbourg, S. Equilibrium traffic assignment on large Virtual Networks: Implementation issues and limits for multi-modal freight transport. *Eur. J. Transp. Infrastruct. Res.* 2006, 6. [CrossRef]
- 83. Liu, S.; Zhou, Y.; Qi, Y.; Chen, Y.; Liu, W.; Xu, H.; Wang, S. Study on the multifunctional spatial-temporal evolution and coupling coordination of cultivated land: A case study of Hebei Province, China. *PLoS ONE* **2024**, *19*, e0306110. [CrossRef] [PubMed]
- 84. Wang, H.; Dong, Y.; Sun, M.; Shi, B.; Ji, H. Dynamic dependence of futures basis between the Chinese and international grains markets. *Econ. Model.* 2024, 130, 106584. [CrossRef]
- 85. Japan Federation of Wood Industry Associations. *Timber Industry, Timber Trade and Timber Legality in China: Trends Survey on the Establishment of Legality Certification for Timber and Timber Products in China;* Japan Federation of Wood Industry Associations (JFWIA): Tokyo, Japan, 2017.
- Ju, H.; Zeng, G.; Zhang, S. Inter-provincial flow and influencing factors of agricultural carbon footprint in China and its policy implication. *Environ. Impact Assess. Rev.* 2024, 105, 107419. [CrossRef]
- Maguire, G. Key Coal Import Hubs in China Perk Up as Economy Reboots. 2022. Available online: https://www.reuters.com/ business/energy/key-coal-import-hubs-china-perk-up-economy-reboots-maguire-2022-12-21/ (accessed on 10 July 2023).
- 88. Wang, S.; Li, B.; Zhao, X.; Hu, Q.; Liu, D. Assessing fossil energy supply security in China using ecological network analysis from a supply chain perspective. *Energy* **2024**, *288*, 129772. [CrossRef]
- 89. Zhang, X.; Yang, X.; He, Q. Multi-scale systemic risk and spillover networks of commodity markets in the bullish and bearish regimes. *N. Am. J. Econ. Financ.* **2022**, *62*, 101766. [CrossRef]
- Sun, X.; Liu, Y.; Guo, S.; Wang, Y.; Zhang, B. Interregional supply chains of Chinese mineral resource requirements. J. Clean. Prod. 2021, 279, 123514. [CrossRef]
- Wu, S.; Lei, Y.; Li, L. Resource distribution, interprovincial trade, and embodied energy: A case study of China. *Adv. Mater. Sci.* Eng. 2015, 2015, 910835. [CrossRef]
- 92. Zhao, Y.; Rogers, S. Tracing China's agrochemical complex. World Dev. 2024, 181, 106675. [CrossRef]
- Zhan, X.; Shao, C.; He, R.; Shi, R. Evolution and efficiency assessment of pesticide and fertiliser inputs to cultivated land in China. Int. J. Environ. Res. Public Health 2021, 18, 3771. [CrossRef] [PubMed]
- 94. Yang, S.; Zhu, Z.; Fu, W.; Hu, S. Tele-connection of embodied carbon emissions from industries in China's trade: A complex network analysis. *J. Environ. Manag.* 2024, 366, 121652. [CrossRef] [PubMed]
- 95. Ren, B.; Li, H.; Wang, X.; Shi, J.; Ma, N.; Qi, Y. The flow of embodied minerals between China's provinces and the world: A nested supply chain network perspective. *Resour. Policy* **2022**, *78*, 102853. [CrossRef]
- Li, X.; Xie, C.; Bao, Z. A multimodal multicommodity network equilibrium model with service capacity and bottleneck congestion for China-Europe containerized freight flows. *Transp. Res. Part E Logist. Transp. Rev.* 2022, 164, 102786. [CrossRef]
- 97. Huang, Y.; Kockelman, K.M. What will autonomous trucking do to US trade flows? Application of the random-utility-based multi-regional input–output model. *Transportation* **2020**, *47*, 2529–2556. [CrossRef]
- Batista, S.F.A.; Leclercq, L. Regional dynamic traffic assignment framework for macroscopic fundamental diagram multi-regions models. *Transp. Sci.* 2019, 53, 1563–1590. [CrossRef]
- Rathod, S.C.; Varia, H.R. Analysis of Truck Trips Generated from Port Infrastructure Based on Trip Length Distribution. In Innovative Research in Transportation Infrastructure: Proceedings of ICIIF 2018; Springer: Berlin/Heidelberg, Germany, 2019; pp. 89–100.

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.